Hitotsubashi Journal of Economics 61 (2020), pp.60-88. © Hitotsubashi University https://doi.org/10.15057/hje.2020004

ENDOGENOUS TIMING IN A MIXED DUOPOLY WITH VERTICALLY RELATED MARKETS*

KI-DONG LEE^{**}

Faculty of Economics & Commerce, Keimyung University Daegu 42601, Republic of Korea kdlee@kmu.ac.kr

Kangsik Choi

Graduate School of International Studies, Pusan National University Busan 46241, Republic of Korea choipnu@pusan.ac.kr

DONGJOON LEE

Faculty of Economics, Osaka Sangyo University Osaka 574–8530, Japan dongjoon@eco.osaka-sandai.ac.jp and Graduate School of Business, Nagoya University of Commerce & Business Nagoya 460–0003, Japan

dongjoon@nucba.ac.jp.

Received March 2019; Accepted February 2020

Abstract

We examine an endogenous timing game in a mixed oligopoly by focusing on the vertical linkages. Our main findings are as follows. First, under discriminatory input pricing, public (private) leadership emerges in a price-setting (quantity-setting) mixed oligopoly. This results contrast with one-tier mixed oligopoly, where a simultaneous-move in Bertrand competition (Bárcena-Ruiz, 2007) or a sequential-move with multiple equilibria in Cournot competition (Pal, 1998) emerges. Second, with downstream Bertrand competition, firms' profit and

^{*} We would like to thank Toshihiro Matsumura, Taiji Furusawa, Dan Sasaki, Akira Ogawa and participants at the seminar held at The University of Tokyo. The authors also wish to thank the editor and anonymous referee of the Journal for their constructive comments. Ki-Dong Lee acknowledges financial support from the Ministry of Education of the Republic of Korea and the National Research Foundation of Korea (NRF-2018S1A5A2A03037352). Kangsik Choi acknowledges that this work was supported by a 2-year Research Grant of Pusan National University. DongJoon Lee acknowledges that this research was supported by JSPS 19H01543, JSPS 1801896, and Osaka Sangyo University.

^{**} Corresponding author

consumer surplus rankings are reversed between uniform and discriminatory input pricing. Finally, banning (allowing) price discrimination on imported inputs is socially desirable under downstream Bertrand (Cournot) competition.

Keywords: endogenous-timing, observable delay game, mixed duopoly, vertically related market, discriminatory input pricing

JEL Classification Codes: D21, H44, L13

I. Introduction

Most of the literature on oligopolistic competition treats the timing of firms' moves (simultaneous or sequential) as exogenous. Firms simultaneously choose quantity (resp. price) in a Cournot (resp. Bertrand) game; by contrast, one firm chooses first, and the other, having observed this, reacts to it in a Stackelberg game. As market outcomes differ depending on whether firms make quantity or price decisions simultaneously or sequentially, determining the timing of moves is important. In their seminal paper, Hamilton and Slutsky (1990) allowed firms to choose the order of moves in an observable delay game and showed that a sequential game occurs in equilibrium if both firms have upward-sloping reaction functions while a simultaneous game occurs if both have downward-sloping reaction functions. The issue of endogenous timing of moves becomes more prominent when applied to a mixed oligopoly in which state-owned public firm and private firms coexist.¹ This is because the differences in the objectives of public firm and private firms — social welfare maximizing and profit maximizing, respectively — may change firm behavior and hence, affect the timing of their moves.

Using an observable delay game, Pal (1998) demonstrated that when firms produce a homogenous good, they decide quantities sequentially in a mixed oligopoly. By contrast, Bárcena-Ruiz (2007) examined the same issue in a price-setting mixed oligopoly and demonstrated that simultaneous moves would occur in equilibrium.² More recently, by introducing various factors influencing market outcomes, the literature on endogenous timing in the mixed oligopoly has become richer and more diverse (Matsumura, 2003; Lu, 2006; Matsumura and Ogawa, 2010; Heywood and Ye, 2009; Bárcena-Ruiz and Garzon 2010; Bárcena-Ruiz and Sedano, 2011; Capuano and De Feo, 2010; Tomaru and Kiyono, 2010; Lee and Xu, 2018; Haraguchi and Hirose, 2018).

However, the existing literature assumes a simple one-tier market, and hence, has devoted

61

¹ According to Hirose and Matsumura (2019), in Japanese financial markets, both public leadership and private leadership emerge in different times. Until the 1980s, public firms had a leading role in the industry. A major change began to take place starting in 2000s, as the Koizumi Cabinet (2001-2006) pursued privatization of public firms under the slogan of "What the private can do must be entrusted to the private". Many of the government-owned firms became privatized and major public institutions were substantially downscaled during this period. That is, the public firms should play a complementary role to private firms, which can be regarded as the private leadership model. However, with the advent of new public financial institutions, public institutions recently began to lead Japanese markets again. In addition, in some developing countries (e.g., China), the public firm is much stronger than the private firms, in which situation the public firm is likely to be the leader in the market (Pi et al., 2018).

² The results of both Pal (1998) and Bárcena-Ruiz (2007) on mixed oligopoly are in sharp contrast to those for a private oligopoly, where quantity competition yields simultaneous-move (Cournot equilibrium) and price competition yields sequential-move choice (Stackelberg equilibrium).

2020] ENDOGENOUS TIMING IN A MIXED DUOPOLY WITH VERTICALLY RELATED MARKETS

62

scant consideration to the impact of vertical linkage in a two-tier market on market outcomes. The vast majority of products are manufactured in multiple production stages of the so-called vertical production chain from raw material to final product. Indeed, in developing countries, coexistence of public and private firms producing final goods from a downstream sector is commonly observed and many manufacturers in the downstream sector depend on imported inputs in the production process of final goods. This could be due to the higher quality of foreign inputs with embedded technology or because certain foreign inputs may not be perfectly substitutable by domestic inputs. Such inputs may include natural resources (e.g., petroleum, natural gas, steel, coal) and transport equipment (e.g., commercial aircraft, mass rapid transit, or railway). For example, in the domestic liquefied natural gas (LNG) market in Korea, the stateowned KOGAS competes with two private firms, POSCO and SK E&S, while the LNG market mostly depends on imports. The Japanese gas market, which is also dependent on imports, can be regarded as a mixed oligopoly consisting of the LP gas service and city-gas service, where city-gas prices and service areas are heavily regulated by the government (Satoh, 2015). As in other developing countries, India's aviation market is a typical mixed oligopoly. India has three full-service airlines: the state-owned Air India and two privately owned airlines, Jet Airways and Tata SIA Airlines. These Indian airlines import more than 70% of their aircraft from Airbus.

In this study, we examine the endogenous order of moves in a mixed oligopoly by focusing on the interaction between upstream and downstream markets. To do this, we consider a vertically-related industry with one foreign upstream manufacturer and two domestic downstream firms — one public and one private — producing differentiated final goods. The upstream foreign monopolist produces an essential intermediate-input that is sold to both domestic firms downstream. The trade between the upstream and downstream firms is conducted through either uniform pricing or discriminatory pricing by the upstream supplier, implying that the upstream monopolist has all the bargaining power in the market.

The main findings of our paper are as follows. First, firms' order of moves depends on whether the foreign upstream supplier uses uniform or discriminatory pricing. Adoption of discriminatory input pricing under downstream Bertrand (resp. Cournot) competition leads to public (resp. private) leadership as a sequential game in equilibrium. These results sharply contrasts with those of a one-tier mixed oligopoly, in which a simultaneous-move equilibrium emerges in a price-setting (Bárcena-Ruiz, 2007) and sequential-move³ multiple equilibria in a quantity-setting (Pal, 1998) mixed oligopoly. The first finding is largely related to the fact that with input price discrimination under downstream Bertrand competition, the upstream monopolist handicaps the private firm by charging higher input price but subsidizes the public firm by charging a lower input price than uniform price (i.e., private firm's relative cost handicap against the public firm). Intuitively, this can be explained via the shape of derived demand for input of each firm.

It is well known that the cost pass-through ratio is lower for the oligopolistic firm than for the firm in the competitive market. In our model, if firms face input price increase, the profit maximizing private firm, which is forced to absorb the cost increase at the expense of its own profits, will pass only a fraction of the rise in input price to the retail price (i.e., low pass-

³ Matsumura and Ogawa (2010) used the concept of risk dominance suggested by Harsanyi and Selten (1988) to show that private leadership is robust when firms compete in terms of quantity in a one-tier mixed duopoly.

through ratio). On the other hand, welfare maximizing public firm, taking into consideration the effects on consumer surplus and private firms' profits, sets its price close to marginal cost, and the increase in input price is simply passed on to the retail price (i.e., high pass-through ratio). As a result, the decrease in final-product sales (input demand under fixed coefficient production function) due to the higher input price is greater for the public firm than for the private firm, which in turn implies that the derived demand curve of the public firm for input is flatter than that of the private firm: i.e., public firm's input demand function is more elastic than the private firm's input demand function. Under discriminatory input pricing, the upstream monopolist offers a lower input price to the public firm to encourage and a higher input price to the private firm to discourage aggressive behavior, thereby earning higher profits than otherwise. In addition, the cost handicap of the private firm is greatest when it is the leader and smallest when it is the follower. The private firm, in order to reduce this cost handicap, strategically chooses to be the follower in the market, which results in public leadership in equilibrium.⁴

Second, the standard rankings for firm's profits and consumer surplus under a one-tier mixed oligopoly are reversed when discriminatory input pricing is adopted in a two-tier mixed oligopoly. Third, upstream supplier's nationality does make a difference for the timing of moves by firms. With a domestic upstream manufacturer, the endogenous order of moves in the downstream mixed oligopoly is consistent between a two-tier and a standard one-tier mixed oligopoly in the downstream market, as long as nonnegative profit constraint is introduced when downstream market is characterized by Cournot competition.

The contributions of this paper are two-fold. First, our results show the importance of interaction between upstream and downstream markets in determining the timing of moves in a mixed oligopoly. We demonstrate that the standard conclusion of the order moves in a one-tier mixed oligopoly is reversed in the two-tier mixed oligopoly when discriminatory input pricing is adopted by upstream firms. Our analysis complements the literature on first- and second-mover advantages (Lee et al., 2017; Amir and Jin, 2001; Amir and Stepanova, 2006; Hoppe, 2000) by analyzing the role of downstream firms in the leader-follower relationship in vertically related markets. Second, our analysis has important implications for competition policy. Price discrimination has long been a contentious issue in competition policy. In many countries, price discrimination is prohibited in the market by government regulations, which forbid dominant firms from charging different buyers different prices for the same product. According to our analysis, the competition mode⁵ (i.e., Cournot or Bertrand) in the downstream market matters in determining whether input price discrimination by the foreign upstream manufacturer should be banded as a matter of policy. Banning price discrimination for imported inputs is desirable

⁴ With downstream Cournot competition, the private firm has a cost handicap only when it is the follower (i.e., public firm leadership). The private firm, by strategically choosing to be the leader in the market, can remove the cost handicap in input prices and increase its profits than otherwise. The public firm takes the strategy to be the follower, because it makes the private firm choose a larger output, which is also welfare-improving. Therefore, private leadership emerges in equilibrium under downstream Cournot competition.

⁵ There are two most common models to describe the strategic interaction between firms in an oligopolistic market: that is, Cournot competition and Bertrand competition. The main difference between the two is whether firms compete in terms of quantity or price. Examples of Cournot competition would be petroleum and natural gas, chemicals, textile, aircraft, shipping containers, and healthcare industry. Examples of Bertrand competition would be airlines, tobacco products, cell phone services, pharmaceutical products, and most of personal service industries. On the other hand, Bloomfield (2018) categorizes 48 industries as Cournot versus Bertrand using three different measures for the mode of competition.

when downstream firms compete à la Bertrand. With the enforcement of uniform pricing for imported inputs, a simultaneous-move equilibrium emerges, leading to higher social welfare and firm profits compared to the public-leadership scenario with discriminatory input pricing. By contrast, government intervention in input pricing is unnecessary with downstream competition à la Cournot. Allowing discriminatory input pricing by the foreign upstream monopolist in effect blocks the possibility of realizing multiple equilibria with public leadership, which is socially less efficient than private leadership.

The remainder of the paper is organized as follows. Section II outlines a simple two-tier mixed duopoly model. Section III examines three types of fixed-timing games in a mixed duopoly: a simultaneous game, public leadership, and private leadership. Section IV compares the market outcomes of these three subgames for both uniform and discriminatory input pricing. Section V analyzes endogenous timing in a mixed-duopoly observable delay game, while Section VI applies the analysis to quantity competition in the downstream market. In Section VII, we examine whether the upstream firm's nationality makes a difference by analyzing a case where the upstream supplier is a domestic firm. Finally, Section VIII provides our concluding remarks.

II. Model

Consider a vertical market structure consisting of an upstream foreign monopolist, denoted by firm M, and two downstream firms, denoted by firm 0 and firm 1, respectively. The upstream firm produces an intermediate-input with zero marginal production cost and exports these intermediate-inputs to the downstream firms domestically. The domestic downstream industry is a mixed oligopoly, where one public firm (firm 0) and one private firm (firm 1) produce differentiated final products solely for sale to domestic consumers. The public firm's objective is to maximize social welfare, whereas the private firm's objective is to maximize its own profits. The production technology that links the upstream and downstream manufacturing is one of fixed proportions: one unit of intermediate input is required to produce one unit of final product.

On the consumption side, there is a continuum of consumers of the same type whose utility function is linear and separate in the numeraire good. The representative consumer maximizes $U-p_0x_0-p_1x_1$, where $x_i \ge 0$, i=0, 1, is the amount of good *i* and p_i its price. The function *U* is assumed to be quadratic, strictly concave and symmetric in x_0 and x_1 : $U=a(x_0+x_1)-\frac{[(x_0)^2+2bx_0x_1+(x_1)^2]}{2}$, where parameter *a* is a positive constant, and $b \in (0, 1)$ denotes the degree of product substitutability, that is, the higher the value of *b*, higher will be the degree of substitutability between products. Given this utility function, the direct and indirect demand functions for good *i* can be derived as follows:

$$x_i = \frac{a(1-b) - p_i + bp_j}{1-b^2}, p_i = a - x_i - bx_j; i, j = 0, 1, i \neq j,$$
(1)

The profit function for each downstream firm is given by

$$\pi_i(\mathbf{p}; w_i) = (p_i - w_i - c_i) x_i(\mathbf{p}), \qquad (2)$$

where $\mathbf{p} \equiv (p_0, p_1)$, w_i is the price of intermediate-input charged to the downstream firm *i*, and c_i is the per-unit production cost of firm *i*. For simplicity, we assume that domestic firms have the same production technology, that is, $c_0 = c_1 = c$. By assuming zero marginal production cost in the intermediate-input production, the profit of the upstream manufacturer (firm M) is given by $\pi_M = \sum_{i=0,1} w_i x_i$. The social welfare for the domestic country is given by

$$W(\mathbf{p}, \mathbf{w}) = \sum_{i=0, 1} \pi_i(\mathbf{p}; w_i) + CS(\mathbf{p})$$
(3)

where $\mathbf{w} \equiv (w_0, w_1)$. The first and second terms of the right-hand side (RHS) in Eq. (3) represent producer surplus and consumer surplus $CS \equiv U - p_0 x_0 - p_1 x_1$, respectively. We make the following assumption throughout the paper.

Assumption: $b \in (0, \bar{b})$, where $\bar{b} = \frac{2}{3}$ for the Bertrand competition and $\bar{b} = \frac{3}{4}$ for the Cournot competition in the downstream market.

Assumption requires that the closeness between products is not too high and guarantees that both domestic firms will produce a positive quantity⁶ of the final-good in all cases under consideration. We use the observable delay game of Hamilton and Slutsky (1990), where firms first choose the timing of their actions. There are two possible time periods for action and each firm chooses its action in only one of the two periods. Our model involves three decisionmaking stages. In stage 1, each of the downstream firms simultaneously chooses whether to set its price in period 1 (T=1) or in period 2 (T=2). There are three possible regimes with respect to the order of firm moves based on the price decision — simultaneous-move, public firm leader, and private firm leader. If the two firms' choices are consistent — public firm (resp. private firm) chooses period 1 and the other chooses period 2, the basic game played is a sequential game with public firm (resp. private firm) as the leader. Otherwise, if both firms choose period 1 or period 2, they receive equilibrium payoffs in a simultaneous-move game. In stage 2, the upstream manufacturer (firm M) sets the price for inputs based on either discriminatory pricing or uniform pricing. In the last stage of the game, firms select their prices knowing when the other firm will make its price choice. Our objective is to solve the subgame perfect Nash equilibrium (SPNE) of this extended game with observable delay using backward induction.

⁶ In discriminatory input price setting, if the products are sufficiently close substitutes, then there might be cases where the upstream monopolist charges an exorbitantly high input price to one of the downstream firms that it becomes inactive and the other downstream firm produces like a monopolist. In our model, as it turns out, if $b \ge \overline{b}$, upstream firm can oust away the private firm and let the public firm produce like a monopolist. We focus on the case where both firms produce positive output to see the strategic interaction of downstream firms in the mixed market (i.e., $b \le \overline{b}$). But this assumption on *b* will be relaxed later and we show that the qualitative nature of our results remains unchanged irrespective of the value of *b*.

III. Mixed Duopoly with Bertrand Competition

We first examine three types of fixed-timing games in a mixed duopoly — simultaneousmove, public leadership, and private leadership. We then examine the endogenous timing game.

1. Simultaneous-Move Game

In the last stage of the game, the first order condition of each downstream firm is given as:

$$\frac{\partial W}{\partial p_0} = (p_0 - c - w_0) \frac{\partial x_0}{\partial p_0} + (p_1 - c - w_1) \frac{\partial x_1}{\partial p_0} = 0 \Leftrightarrow R_0(p_1; \mathbf{w}) = (c + w_0) + b(p_1 - c - w_1), \quad (4.1)$$
$$\frac{\partial \pi_1}{\partial p_1} = x_1 + (p_1 - c - w_1) \frac{\partial x_1}{\partial p_1} = 0 \Leftrightarrow R_1(p_0; w_1) = \frac{1}{2} [a(1 - b) + bp_0 + c + w_1], \quad (4.2)$$

where R_i is the reaction function of firm *i*. In Eq. (4.1), $(p_1 - (c+w_1))\frac{\partial x_1}{\partial p_0}$ (>0) represents a profit-driven effect for the private firm. When the public firm chooses its price level, it takes into consideration the fact that an increase in its own price over its marginal cost will increase the private firm's output (and thus, profits too) via substitution effect, and hence, will increase social welfare. Therefore, the public firm under Bertrand competition⁷ will set its price strictly higher than the marginal cost; $[p_0]_{Bertrand} = (c+w_0) + r(p_1 - c - w_1) > (c+w_0)$. The intersection of the two reaction curves gives the equilibrium values of retail prices and quantities at this stage of the game (the superscript "S" denotes the simultaneous-move):

$$p_{0}^{s}(\mathbf{w}) = \frac{ab(1-b)+2(c+w_{0})-b(c+w_{1})}{2-b^{2}}, p_{1}^{s}(\mathbf{w}) = \frac{a(1-b)+(1-b^{2})(c+w_{1})+b(c+w_{0})}{2-b^{2}},$$
(5.1)

$$x_{0}^{s}(\mathbf{w}) = \frac{(a-c-w_{0})-b(a-c-w_{1})}{1-b^{2}}, x_{1}^{s}(\mathbf{w}) = \frac{(a-c-w_{1})-b(a-c-w_{0})}{(1-b^{2})(2-b^{2})},$$
 (5.2)

where $\left|\frac{\partial x_0^s}{\partial w_0}\right| = \frac{1}{1-b^2} > \frac{1}{(1-b^2)(2-b^2)} = \left|\frac{\partial x_1^s}{\partial w_1}\right|$, suggests that the derived demand for inputs of firm 0 is more elastic than the derived demand for the inputs of firm 1. The maximization problem of firm M in stage 2 is $\operatorname{Max}_{w_0, w_1} \pi_M^s(\mathbf{w}) \left(\equiv \sum_{i=0, 1} w_i x_i^s(\mathbf{w})\right)$. By solving $\frac{\partial \pi_M^s}{\partial w_0} = \frac{\partial \pi_M^s}{\partial w_1} = 0$ simultaneously, we obtain equilibrium input prices in the downstream simultaneous-move game as follows (the superscript "*" denotes the equilibrium under discriminatory input pricing):

$$\frac{(4-b-b^2)\Gamma}{8-5b^2+b^4} = w_0^{5^*} < w_1^{5^*} = \frac{(2-b^2)(2+b-b^2)\Gamma}{8-5b^2+b^4},$$
(6)

⁷ Under Cournot competition, the welfare-maximizing public firm chooses its own output, taking the rival's output as given. Therefore, the public firm behaves like a welfare maximizing monopolist, and will produce its output based on the traditional marginal cost pricing condition, that is, $[p_0]_{Cournot} = c + w_0$.

				HITOTSUBASHI	JOURNAL O	F ECONOMICS	
NO	d	$b \in \left[\frac{2}{3}, 1\right)$	$w_{0}^{p,r^*} = \frac{\Gamma}{2}$ $w_{1}^{p,r^*} = \frac{(2-b)\Gamma}{2}$	$p_{0^{m^*}}^{p_{m^*}} = \frac{a+c}{2}$ $p_{1^{m^*}}^{p_{m^*}} = \frac{a(2-b)+bc}{2}$	$x_{0}^{p_{2}} = \frac{a+c}{2}$ $x_{1}^{p_{3}} = 0$	$\pi_0^{Pa*} = 0$ $\pi_1^{Pa*} = 0$	$CS^{p_{11}} = \frac{\Gamma^2}{8}$ $W^{p_{12}} = \frac{\Gamma^2}{8}$
NSTREAM PRICE COMPETITIC	Private Leadershi	$b \in \left(0, \frac{2}{3}\right)$	$\begin{split} w_{0}^{P,r*} = & \frac{(1-b)(4+3b)\Gamma}{8-9b^2} \\ & w_{1}^{P,r*} = & \frac{2(1-b)(2+3b)\Gamma}{8-9b^2} \end{split}$	$p_{p^{p,*}}^{p,*} = \frac{(4+b-6b^2)a + (4-b-3b^2)c}{8-9b^2}$ $p_{p^{p,*}}^{p,*} = \frac{(6-b-6b^2)a + (2+b-3b^2)c}{8-9b^2}$	$x_{0}^{p,t} = \frac{(4-3b)(a-c)}{8-9b^2}$ $x_{1}^{p,t} = \frac{(2-3b)\Gamma}{8-9b^2}$	$\begin{aligned} \pi_{0}^{p,n^{*}} &= bx_{0}^{p,n^{*}} x_{1}^{p,n^{*}} \\ &= \frac{b(4-3b)(2-3b)\Gamma^{2}}{(8-9b^{2})^{2}} \\ \pi_{1}^{p,n^{*}} &= (x_{1}^{p,n^{*}})^{2} \\ &= \frac{(2-3b)^{2}\Gamma^{2}}{(8-9b^{2})^{2}} \end{aligned}$	$CS^{Pot*} = \frac{(1-b)(10-9b^2)\Gamma^2}{(8-9b^2)^2}$ $W^{Pot*} = \frac{2(1-b)(7-9b^2)\Gamma^2}{(8-9b^2)^2}$
JILIBRIUMS UNDER EACH REGIME: DOW		Public Leadership	$\begin{split} w_{0}^{p_{diff}} &= \frac{(16-2b-18b^2+b^3+5b^4)\Gamma}{32-41b^2+13b^4} \\ w_{1}^{p_{diff}} &= \frac{(16+4b-23b^2-4b^3+8b^4+b^5)\Gamma}{32-41b^2+13b^4} \end{split}$	$p_{0}^{p_{ms}} = \frac{\left[\left[16 + 2b - 23b^{2} - b^{3} + 8b^{4} \right)a}{\left[+ \left(16 - 2b - 18b^{2} + b^{3} + 5b^{4} \right)c} \right]}{\left(32 - 41b^{2} + 13b^{4} \right)}$ $p_{0}^{p_{ms}} = \frac{\left[\left(24 - 6b - 31b^{2} + 7b^{3} + 10b^{4} - 2b^{5} \right)a}{\left(32 - 41b^{2} + 13b^{4} \right)} \right]}{\left(32 - 41b^{2} + 13b^{4} \right)}$	$x_{0^{\text{dot}}}^{\text{hose}} = \frac{(2-b^2)(8+3b-5b^2-2b^3)\Gamma}{(1+b)(32-41b^2+13b^4)}$ $x_{1^{\text{hose}}}^{\text{hose}} = \frac{(1-b)(2-b^2)(4+3b)\Gamma}{(1+b)(32-41b^2+13b^4)}$	$\begin{split} \pi_{0}^{\mu_{0}\mu^{*}} &= \frac{b(1-b^{2})}{(2-b^{2})} \pi_{0}^{p_{0}\mu^{*}} r_{1}^{\mu_{0}\nu^{*}} \\ &= \frac{b(1-b^{2})^{2}(4+3b)(2-b^{2})(8+3b-5b^{2}-2b^{3})\Gamma^{2}}{(1+b)(32-41b^{2}+13b^{4})^{2}} \\ \pi_{1}^{\mu_{0}\mu^{*}} &= (1-b^{2})(x^{p_{0}}\mu^{*})^{2} \\ &= \frac{(1-b)^{3}(8+6b-4b^{2}-3b^{3})^{2}\Gamma^{2}}{(1+b)(32-41b^{2}+13b^{4})^{2}} \end{split}$	$CS^{p_{\rm M}*} = \frac{(2-b^2)^2 (40+12b-55b^2-20b^3+19b^4+8b^5)\Gamma^2}{(1+b)(32-41b^2+13b^4)^2} \\ W^{p_{\rm M}*} = \frac{(224+16b-560b^2-50b^3+522b^4+57b^5)}{(-215b^6-28b^2+33b^8+5b^9)} \\ M^{p_{\rm M}*} = \frac{(1+b)(32-41b^2+13b^4)^2}{(1+b)(32-41b^2+13b^4)^2} $
Table 1. Market Equ		Simultaneous Move	$w_0^{s^*} = \frac{(4-b-b^2)\Gamma}{8-5b^2+b^4}$ $w_1^{s^*} = \frac{(2-b^2)(2+b-b^2)\Gamma}{8-5b^2+b^4}$	$p_{0}^{s} = \frac{(4+b-4b^{2}+b^{4})a + (4-b-b^{2})c}{(8-5b^{2}+b^{4})}$ $p_{1}^{s} = \frac{(6-b-4b^{2}+b^{4})a + (2+b-b^{2})c}{(8-5b^{2}+b^{4})}$	$x_{1}^{5} = \frac{(4+b-b^{2})\Gamma}{(1+b)(8-5b^{2}+b^{4})}$ $x_{1}^{5} = \frac{(2-b-b^{2})\Gamma}{(2-b-b^{2})\Gamma}$	$\begin{aligned} \pi_1^{s_1} &= b(1-b^2) \chi_1^{s_1} \chi_1^{s_1} \\ &= \frac{b(4+b-b^2)(2-3b+b^3) \Gamma^2}{(1+b)(8-5b^2+b^4)^2} \\ &\pi_1^{s_1} &= (1-b^2)(\chi_1^{s_1})^2 \\ &= \frac{(1-b^2)(\chi_1^{s_1})^2}{(1+b)(8-5b^2+b^4)^2} \end{aligned}$	$CS^{s} = \frac{(10 - 7b^2 + b^4)\Gamma^2}{(1 + b)(8 - 5b^2 + b^4)^2}$ $W^{s} = \frac{(14 - 16b^2 + 7b^4 - b^6)\Gamma^2}{(1 + b)(8 - 5b^2 + b^4)^2}$
			Input Price	Output Price	Quantities	Profit	Welfare

Note: $\Gamma \equiv a - c$

[June

where $\Gamma \equiv a - c$. Substituting $w_0^{S^*}$ and $w_1^{S^*}$ into market variables, we obtain the equilibrium values for those variables as summarized in Table 1.

2. Sequential-move Game: Public Firm Leadership

In this case, the public firm sets price level p_0 to maximize social welfare anticipating the private firm's response. The maximization problem of the public firm is $\operatorname{Max}_{p_0}W^{P_{ub}}(p_0;\mathbf{w})$ ($\equiv W(p_0,R_1(p_0,w_1);\mathbf{w})$), where superscript '*Pub*' denotes the public leadership and $R_1(p_0;w_1)$ is given by Eq. (4.2). Solving the first order condition of welfare maximization, we obtain

$$p_{0}^{Pub}(\mathbf{w}) = \frac{ab(1-b)+2(2-b^{2})(c+w_{0})-b(c+w_{1})}{4-3b^{2}},$$

$$p_{1}^{Pub}(\mathbf{w}) = \frac{(2-2b-b^{2}+b^{3})a+b(2-b^{2})(c+w_{0})+2(1-b^{2})(c+w_{1})}{4-3b^{2}},$$
(7.1)

$$x_{0}^{Pub}(\mathbf{w}) = \frac{(2-b^{2})^{2}(a-c-w_{0})-b(3-2b^{2})(a-c-w_{1})}{(4-3b^{2})(1-b^{2})},$$

$$x_{1}^{Pub}(\mathbf{w}) = \frac{(2-b^{2})\{(a-c-w_{1})-b(a-c-w_{0})\}}{(4-3b^{2})(1-b^{2})}.$$
(7.2)

When compared to the simultaneous case, $p_i^{Pub}(\mathbf{w}) \le p_i^S(\mathbf{w})(i=0, 1)$ holds. This implies that the welfare-maximizing public firm as the price leader can increase social welfare by setting a price lower than the price in the simultaneous-move game. The follower (private firm) will lower its price too because prices are strategic complements. In stage 2, the maximization problem of firm M is $\operatorname{Max}_{w_0,w_1}\sum_{i=0,1} w_i x_i^{Pub}(\mathbf{w})$, where $x_i^{Pub}(\mathbf{w})$ is given by Eq. (7.2). By solving $\frac{\partial \pi_M^{Pub}}{\partial w_0} = \frac{\partial \pi_M^{Pub}}{\partial w_1} = 0$ simultaneously, we obtain the equilibrium input prices under public leadership as follows:

$$\frac{(16-2b-18b^2+b^3+5b^4)\Gamma}{32-41b^2+13b^4} = w_0^{Pub^*} < w_1^{Pub^*} = \frac{(16+4b-23b^2-4b^3+8b^4+b^5)\Gamma}{32-41b^2+13b^4}, \quad (8)$$

Equilibrium input prices $w_i^{Pub^*}$ will lead to the equilibrium market outcomes under downstream public leadership in a sequential-move game as summarized in Table 1.

3. Sequential-move Game: Private Firm Leadership

In this game, the maximization problem of the private firm is $\operatorname{Max}_{p_1} \pi_1^{Pri}(p_1; \mathbf{w})$ $(\equiv \pi_1(p_1, R_0(p_1; \mathbf{w}); w_1))$, where $R_0(p_1; \mathbf{w})$ is given in Eq. (4.1). Solving $\frac{\partial \pi_1^{Pri}}{\partial p_1} = 0$ gives the following (superscript "*Pri*" denotes private leadership):

$$p_{0}^{Pri}(\mathbf{w}) = \frac{ab(1-b) + (2-b^{2})(c+w_{0}) - b(c+w_{1})}{2(1-b^{2})},$$

$$p_{1}^{Pri}(\mathbf{w}) = \frac{a(1-b) + (1-2b^{2})(c+w_{1}) + b(c+w_{0})}{2(1-b^{2})},$$
(9.1)

HITOTSUBASHI JOURNAL OF ECONOMICS

$$x_{0}^{Pri}(\mathbf{w}) = \frac{(a-c-w_{0})-b(a-c-w_{1})}{1-b^{2}}, x_{1}^{Pri}(\mathbf{w}) = \frac{(a-c-w_{1})-b(a-c-w_{0})}{2(1-b^{2})},$$
(9.2)

In contrast to the simultaneous case, we get $p_i^{Pri}(\mathbf{w}) > p_i^S(\mathbf{w})$, implying that the private firm tends to set a higher price than that in the simultaneous-move game when it is the price leader. This reflects the fact that that private firms, profit maximizers, aim to reduce market competition because they are profits maximizers. As in the previous case, the maximization problem of the upstream manufacturer is $\operatorname{Max}_{w_0,w_1} \sum_{i=0,1} w_i x_i^{Pri}(\mathbf{w})$, where $x_i^{Pri}(\mathbf{w})$ is given by Eq. (9.2). By solving $\frac{\partial \pi_M^{Pri}}{\partial w_0} = \frac{\partial \pi_M^{Pri}}{\partial w_1} = 0$ simultaneously, we obtain equilibrium input prices under private leadership as follows:

$$\frac{(1-b)(4+3b)\Gamma}{8-9b^2} = w_0^{Pri^*} < w_1^{Pri^*} = \frac{2(1-b)(2+3b)\Gamma}{8-9b^2},$$
(10)

Using $w_0^{p_{rl}*}$ and $w_1^{p_{rl}*}$, we obtain the equilibrium market outcomes under downstream private leadership, as shown in Table 1. Here, it should be noted that the equilibrium output of the private firm is positive $\left(x_1^{p_{rl}*}=\frac{(2-3b)\Gamma}{8-9b^2}>0\right)$ only when $b<\frac{2}{3}$, implying that if the goods are close substitutes (i.e., $b \in \left[\frac{2}{3}, 1\right)$), then only the public firm will serve the market because $x_1^{p_{rl}*}=0$.

The following lemma regarding equilibrium input prices (Eqs. 6, 8, and 10) is immediate.

Lemma 1: (1)
$$w_1^{I*} > \frac{a-c}{2} > w_0^{I*}$$
 for $I = S, Pub, Pri$,
(2) $w_1^{Pri*} > w_1^{S*} > w_1^{Pub*}$ and $\Delta w^{Pri*} > \Delta w^{S*} > \Delta w^{Pub*} > 0$, where $\Delta w^{I*} \equiv w_1^{I*} - w_0^{I*}$.

Proof: The proof is easily obtained with simple calculations based on Eqs. (6), (8), and (10).

Lemma 1(1) implies that the upstream supplier charges the private firm a higher input price and the public firm a lower price compared to $\frac{a-c_8}{2}$; that is, the price-discriminator (upstream supplier) handicaps the private firm while subsidizing the public firm through input pricing. This is explained by the fact that the derived demand for the inputs of firm 0 is more sensitive than that of firm 1 to changes in its own input price and in the input price charged to the rival firm (private firm); that is, $\left|\frac{\partial x_0^i}{\partial w_0^i}\right| > \left|\frac{\partial x_1^i}{\partial w_1^i}\right|$ and $\left|\frac{\partial x_0^i}{\partial w_1^i}\right| > \left|\frac{\partial x_1^i}{\partial w_0^i}\right|$ for I=S,Pri,Pub. In this case, the upstream monopolist wishes the downstream public firm to behave aggressively while the private firm to behave less aggressively in the market. The upstream monopolist offers a lower input price to the public firm to encourage and a higher input price to the private firm to discourage aggressive behavior, thereby earning higher profits than otherwise.

The cost handicap of the private firm is greatest when it is the price leader and smallest

69

[June

 $[\]frac{a-c}{2}$ is the input price level under uniform input pricing, which will be discussed later immediately.

ENDOGENOUS TIMING IN A MIXED DUOPOLY WITH VERTICALLY RELATED MARKETS

when it is the follower (Lemma 1(2)). The derived demand for the inputs of the private firm, when it is the price leader, is inelastic with respect to the input price (w_1) , in contrast to the simultaneous-move case; that is, $\left|\frac{\partial x_1^{Pri}}{\partial w_1}\right| = \frac{1}{2(1-b^2)} < \frac{1}{(1-b^2)(2-b^2)} = \left|\frac{\partial x_1^s}{\partial w_1}\right|^9$ This leads to $w_1^{Pri^*} > w_1^{S^*}$ because w_1 has an identical effect on the public firm's output in both private leadership and the simultaneous-move case $\left(\left|\frac{\partial x_1^{0ri}}{\partial w_1}\right| = \left|\frac{\partial x_0^s}{\partial w_1}\right| = \frac{b}{(1-b^2)}\right)$. As in private leadership, the derived demand for the inputs of the private firm, when the public firm is the price leader, is inelastic with respect to w_1 , in contrast to the simultaneous-move case $\left(i.e., \left|\frac{\partial x_1^{Pri}}{\partial w_1}\right| = \frac{2-b^2}{(1-b^2)(4-3b^2)} < \frac{1}{(1-b^2)(2-b^2)} = \left|\frac{\partial x_1^s}{\partial w_1}\right|$, which tends to $w_1^{S^*} < w_1^{Pub^*}$. By contrast, the public firm's demand for inputs is less sensitive to a change in w_1 with the public firm as the price leader compared to simultaneous-move case $\left(i.e., \left|\frac{\partial x_1^{0}}{\partial w_1}\right| = \frac{b(3-2b^2)}{(1-b^2)(4-3b^2)} < \frac{b}{(1-b^2)} = \left|\frac{\partial x_0^s}{\partial w_1}\right|$, which tends to $w_1^{S^*} > w_1^{Pub^*}$. As the latter dominates the former, $w_1^{S^*} > w_1^{Pub^*}$ holds.

IV. Comparisons

1. Uniform Input Pricing

2020]

Before we compare the market outcomes in the three regimes, we consider the benchmark case where the upstream manufacturer adopts uniform pricing.¹⁰ In this case, the upstream manufacturer will maximize the following expression to determine the input price: $\operatorname{Max}_w \sum x_i^l(w, w)$, where $x_i^l(w, w)$ is given by Eqs. (5.2), (7.2), and (9.2) with $w_0 = w_1 = w$. Solving $\frac{d\pi_M^l(w)}{dw} = 0$ for w yields $w^{s+} = w^{Pub+} = w^{Pri+} = \frac{\Gamma}{2}$, where "+" denotes equilibrium in uniform pricing. The input prices imposed on downstream firms are the same irrespective of the state of the firm. This implies that, as long as uniform pricing is adopted, the presence of a vertically related upstream sector does not substantially affect the endogenous timing of firms' moves in a mixed oligopoly. The following lemma confirms the results in Bárcena-Ruiz (2007), who examined the same issue in a one-tier mixed oligopoly.

Lemma 2: The following holds in equilibrium under uniform input pricing:

⁹ Under price competition, the private firm, as Stackelberg leader, chooses a higher price to increase its profits than in the simultaneous-move game. Therefore, private leadership in the price competition, compared to simultaneousmove game, reduces market competition, which leads to lower cost pass-through ratio of the private firm in the private leadership. Since input price is the cost of products, this implies that the derived demand for input of the private firm in the private-leadership is less elastic than in the simultaneous-move game.

¹⁰ From the perspective of profit maximization, the upstream supplier would clearly prefer price discrimination to uniform pricing in the presence of asymmetric downstream firms. However, government regulations sometimes prohibit price discrimination in the input market; therefore, the upstream monopolist must charge the same price for all downstream firms.

$$p_1^{Pri+} > p_1^{S+} > p_1^{Pub+}, p_0^{Pri+} > p_0^{S+} > p_0^{Pub+}, \pi_1^{Pri+} > \pi_1^{S+} > \pi_1^{Pub+}, \pi_0^{Pri+} > \pi_0^{S+} > \pi_0^{S+} > \pi_0^{Pub+}, CS^{Pub+} > CS^{S+} > CS^{Pri+}, \text{ and } W^{Pub+} > W^{S+} > W^{Pri+}.$$

Proof: Given in Appendix 1.

Considering that private firms would want to reduce market competition (i.e., raise their prices) as profit maximizers while public firm would want to raise market competition (i.e., reduce its price) as a welfare maximizer, Lemma 2 is straightforward. When the private firm is the price leader, it will set a price higher than the price in the simultaneous-move because it knows that the public firm (the follower) will also raise its price because of strategic complementarity in prices (i.e., $p_1^{Pri+} > p_1^{S+}$ and $p_0^{Pri+} > p_0^{S+}$). By contrast, when the public firm is the price leader, it will set a price lower than the price in the simultaneous case because the public firm knows that the private firm will also lower its price (i.e., $p_0^{Pub+} < p_0^{S+}$ and $p_1^{Pub+} < p_1^{S+}$). As a result, market competition is stronger (resp. weaker) when the public (resp. private) firm is a leader than otherwise. Since the two firms set higher (resp. lower) prices under private (resp. public) leadership than under simultaneous-move game, the profits of the firms will be higher (resp. lower) and consumer surplus will be lower (resp. higher), and welfare will be lower (resp. higher) under private (resp. public) leadership than when firms' decisions are set simultaneously (i.e., $\pi_i^{Pri+} > \pi_i^{S+} > \pi_i^{Pub+}$, $CS^{Pub+} > CS^{S+} > CS^{Pri+}$, and $W^{Pub+} > W^{S+} > W^{Pri+}$).

2. Discriminatory Input Pricing

By comparing the market outcomes under discriminatory input pricing, we obtain the following proposition (superscript "*" denotes equilibrium in discriminatory input pricing).

Lemma 3: $x_0^{I^*} - x_1^{I^*} > x_0^{I^+} - x_1^{I^+} > 0$ for I = S, Pub, Pri.

Proof: From Eqs. (5.2), (7.2) and (9.2), we have $x_0^s - x_1^s = \frac{(1-b)^2(a-c-w_0^s) + (1+b-b^2)\Delta w^s}{(1-b)(2-b^2)}$, $x_0^{p_{ub}} - x_1^{p_{ub}} = \frac{(2-3b+b^3)(a-c-w_0^{p_{ub}}) + (2+b-b^2)\Delta w^{p_{ub}}}{(1-b)(4-3b^2)}$, and $x_0^{p_{ri}} - x_1^{p_{ri}} = \frac{(1-b)(a-c-w_0^{p_{ri}}) + (1+2b)\Delta w^{p_{ri}}}{2(1-b^2)}$. Since $\Delta w^I = 0$ under uniform pricing, we have $x_0^{I+} - x_1^{I+} > 0$. In addition, since $x_0^I - x_1^I$ is a positive function in Δw^I and a negative function in w_0^I , it follows that $x_0^{I*} - x_1^{I*} > x_0^{I+} - x_1^{I+}$ because $\Delta w^{I*} > \Delta w^{I+} (=0)$ and $w_0^{I*} < w^+ \left(= \frac{a-c}{2} \right)$.

Lemma 3 implies that since the upstream supplier handicaps the private firm and subsidizes the public firm in discriminatory input pricing, production shifts from the private firm to the public firm. Therefore, the production gap between the public and private firms widens in comparison with uniform input pricing. From Table 1, the following proposition is immediate.

Proposition 1: (1) $p_1^{Pri^*} > p_1^{S^*} > p_1^{Pub^*}$, (2) $x_1^{Pub^*} > x_1^{S^*} > x_1^{Pri^*}$ and $x_0^{Pri^*} - x_1^{Pri^*} > x_0^{S^*} - x_1^{S^*} > x_0^{Pub^*} - x_1^{Pub^*} > 0$.

(3)
$$x_0^{Pub^*} + x_1^{Pub^*} > x_0^{S^*} + x_1^{S^*} > x_0^{Pri^*} + x_1^{Pri^*}$$
, (4) $\pi_1^{Pub^*} > \pi_1^{S^*} > \pi_1^{Pri^*}$.

Proof: See Appendix 2.

In Proposition 1, $p_1^{Pri^*} > p_1^{S^*} > p_1^{Pub^*}$ is straightforward. Note that $p_1^{Pri^+} > p_1^{S^+} > p_1^{Pub^+}$ holds under uniform input pricing. Since input prices are passed through to the output prices, $w_1^{Pri^*} > w_1^{S^*} > w_1^{Pub^*}$ (Lemma 1(2)) implies $p_1^{Pri^*} > p_1^{S^*} > p_1^{Pub^*}$. The cost handicap of the private firm against the public firm is largest when the private firm is the price leader and smallest when it is the follower ($\Delta w^{Pri^*} > \Delta w^{S^*} > \Delta w^{Pub^*}$), which leads to $x_1^{Pri^*} < x_1^{S^*} < x_1^{Pub^*}$ and $\pi_1^{Pub^*} > \pi_1^{S^*} > \pi_1^{Pri^*}$. In addition, aggregate output is largest when the public firm is the leader, lower in the simultaneous case, and, lowest when it is the follower. Compared to the ranking under uniform pricing, $\pi_1^{Pub^*} > \pi_1^{S^*} > \pi_1^{Pri^*}$ seems somewhat paradoxical. Unlike in uniform input pricing, the private firm's cost disadvantage (i.e., handicap in input price) against the public firm is largest when it is the price leader and lowest when it is the follower. As a result, the profit ranking of the private firm under discriminatory input pricing is the opposite of the input price ranking, $w_1^{Pri^*} > w_1^{S^*} > w_1^{Pub^*}$.

Next, we turn to the price ranking of the public firm. By comparing the equilibrium price of good 0 under different regimes presented in Table 1, we obtain the following lemma:

Lemma 4: Suppose discriminatory pricing in the upstream sector; the following results hold:

$$\begin{cases}
p_0^{Pri^*} > p_0^{S^*} > p_0^{Pub^*} & \text{if } b \in (0, \ 0.447) \\
p_0^{S^*} > p_0^{Pri^*} > p_0^{Pub^*} & \text{if } b \in (0.447, \ 0.582) \\
p_0^{S^*} > p_0^{Pub^*} > p_0^{Pri^*} & \text{if } b \in \left(0.582, \frac{2}{3}\right)
\end{cases}$$

Proof: See Appendix 3.

If b is sufficiently low (i.e., b < 0.447), the price ranking of the public firm under discriminatory input pricing $(p_0^{p_{ri}^*} > p_0^{S^*} > p_0^{p_{ub}^*})$ is consistent with that under uniform pricing. This is straightforward, considering that if the goods are sufficiently independent, then the gap in equilibrium input prices across different regimes almost vanishes.¹¹ However, as b increases, $p_0^{p_{ub}^*}$ becomes relatively high while $p_0^{p_{ri}^*}$ becomes relatively low, resulting in a reversal of the public firm's price ranking.

This is largely related to the pricing behavior of the public firm as a welfare maximizer. Recall that the public firm sets its price by adding a profit-driven effect for the private firm to its marginal cost (i.e., $[p_0]_{Bertrand} = (c+w_0) + b(p_1 - c - w_1)$, see Eq. (4.1)). This profit driven effect, $b(p_1 - c - w_1)$, is stronger as x_1 (i.e., price-cost margin) increases. In Proposition 1(2), we have $x_1^{Pub^*} > x_1^{S^*} > x_1^{Pri^*}$. As goods become closer substitutes (an increase in b), the leaderfollower relationship exerts a greater influence on the market outcome, expanding the output gap across different regimes. This results in a relative decrease in $x_1^{Pri^*}$ and a relative increase in $x_1^{Pub^*}$, which in turn leads to a relative decrease in $p_0^{Pri^*}$ and a relative increase in $p_0^{Pub^*}$.

As for the consumer surplus and total surplus, the following proposition can be obtained.

2020]

¹¹ If *b* approaches zero, we find that $\lim_{b\to 0} w_0^{S^*} = \lim_{b\to 0} w_0^{Put^*} = \lim_{b\to 0} w_0^{Pub^*} = \frac{a-c}{2}$.

Proposition 2: Suppose discriminatory pricing in the upstream sector; the following holds true: $\begin{bmatrix} CS^{Pub^*} > CS^{S^*} > CS^{Pri^*} & \text{if } b \in (0, 0.587) \end{bmatrix}$

(1)
$$\begin{cases} CS^{Pub^*} > CS^{Pri^*} > CS^{S^*} & \text{if } b \in (0.587, 0.641), \\ CS^{Pri^*} > CS^{Pub^*} > CS^{S^*} & \text{if } b \in \left(0.641, \frac{2}{3}\right) \end{cases}$$
, (2) $W^{Pub^*} > W^{S^*} > W^{Pri^*}$.

Proof: See Appendix 4.

The consumer surplus ranking is almost the opposite of the public firm's output price ranking in Lemma 4. Since $x_0^{t*} > x_1^{t*}$ in Proposition 1(2), p_0^{t*} has a greater impact on consumer surplus than p_1^{t*} in all cases. Whereas the consumer surplus ranking varies depending on b, the social welfare ranking remains constant and coincides with that under uniform input pricing. This can be explained as follows. Note that $W = f(x_0+x_1) + d(x_0-x_1) - \pi_M$, where $f(\cdot) = (a-c)(x_0+x_1) - \frac{1+b}{4}(x_0+x_1)^2$, $d(\cdot) = -\frac{1-b}{4}(x_0-x_1)^2$, and $\pi_M = \sum_{i=0,1} w_i x_i$. Here, $x_0^{pub*} + x_1^{pub*} > x_0^{5*} + x_1^{5*} > x_0^{pri*} + x_1^{pri*}$ and $x_0^{pub*} - x_1^{pub*} > x_0^{5*} - x_1^{5*} > x_0^{pri*} - x_1^{pri*}$ in Proposition 1 (2) makes $W^{pub*} > W^{5*} > W^{Pri*}$ more likely since $f'(\cdot) > 0$ and $d'(\cdot) < 0$ while $\pi_M^{Pub*} > \pi_M^{5*} > \pi_M^{pri*12}$ makes it less likely. It emerges that the former effect dominates the latter, resulting in the standard welfare ranking.

What is noteworthy here is that the standard leader-follower rankings under uniform input pricing are reversed for consumer surplus if *b* is sufficiently high. In Proposition 2 (1), if $b \in \left(0.641, \frac{2}{3}\right)$, then $CS^{Pub^*} > CS^{Pub^*}$, which contrasts with the conventional view that public leadership in price is more beneficial to the consumers than private leadership. Despite the reversal in the leader-follower rankings for the consumer, the welfare ranking remains the same in both discriminatory pricing and uniform input pricing.

V. Endogenous Timing Game

1. CASE 1: $b \in \left(0, \frac{2}{3}\right)$

We now discuss the first-stage choice in an endogenous timing game between two downstream firms that produce positive quantities in the market. Table 2 provides the payoff matrix of the observable delay game in a mixed oligopoly. Each firm *i* simultaneously chooses whether to move early (T=1) or late (T=2). First, we examine the case of uniform input pricing in the upstream sector. From Lemma 2, the following lemma is immediate.

Lemma 5: Suppose that the downstream firms compete in price. (1) Under uniform input pricing, a "simultaneous-move" occurs, where both firms choose their prices in period 1. (2) However, the equilibrium outcome is inefficient.

¹² Notice that $\pi_M^{I^*} = \sum_{i=0,1} w_i^{I^*} x_i^{I^*} = \frac{a-c}{2} \sum_{i=0,1} x_i^{I^*}$ holds under discriminatory input pricing. Considering this, $\sum x_i^{Pub^*} > \sum x_i^{S^*} > \sum x_i^{Pri^*}$ in Proposition 1(3) implies $\pi_M^{Pub^*} > \pi_M^{S^*} > \pi_M^{Pri^*}$.

2020]

Proof: In Lemma 2, $W^{Pub^+} > W^{S^+} > W^{Pri^+}$ for firm 0 and $\pi_1^{Pri^+} > \pi_1^{S^+} > \pi_1^{Pub^+}$ for firm 1 imply setting prices in T=1 is the dominant strategy for both firms. This leads to (1). Equilibrium payoff $(W^{S^+}, \pi_1^{S^+})$ is not efficient because $W^{Pub^+} > W^{S^+}$ and $\pi_1^{Pri^+} > \pi_1^{S^+}$ in Lemma 2. This leads to (2).

Firm 1 Firm 0	T=1	<i>T</i> =2
T=1	$W^{S\Psi}, \pi_1^{S\Psi}$	$W^{Pub\Psi},\pi_1^{Pub\Psi}$
<i>T</i> =2	$W^{{\scriptscriptstyle Pri}{\scriptstyle {\scriptstyle {ar ar \Psi}}}},\pi_{\scriptscriptstyle 1}^{{\scriptscriptstyle Pri}{\scriptstyle {\scriptstyle {ar ar \Psi}}}}$	$W^{S^{ alpha}}, \pi_1^{S^{ alpha}}$

 TABLE 2: PAYOFF MATRIX IN A MIXED DUOPOLY

Note: $\Psi = +$ for uniform pricing while $\Psi = *$ for discriminatory pricing.

Lemma 5 coincides the results obtained by Barcena-Ruiz (2007) who examined a similar issue in a standard one-tier mixed oligopoly. This implies that the presence of vertically related markets does not substantially impact the endogenous order of moves in a mixed oligopoly as long as uniform input pricing is adopted. We now examine the order of firms' moves under discriminatory input pricing. The following proposition is obtained.

Proposition 3: If the upstream monopolist adopts discriminatory input pricing, then "public leadership" is the Nash equilibrium outcome, which is Pareto efficient.

Proof: $W^{Pub^*} > W^{S^*} > W^{Pri^*}$ (Proposition 2(2)) implies T=1 is the dominant strategy for firm 0, while $\pi_1^{Pub^*} > \pi_1^{S^*} > \pi_1^{Pri^*}$ (Proposition 1(4)) implies T=2 is the dominant strategy for firm 1. Therefore, public leadership is SPNE. In addition, since $\pi_1^{Pub^*} > \pi_1^{S^*} > \pi_1^{Pri^*}$ and $W^{Pub^*} > W^{S^*} > W^{Pri^*}$, the equilibrium payoff $(W^{Pub^*}, \pi_1^{Pub^*})$ is an efficient one.

The intuition behind Proposition 3 is as follows. As in the uniform pricing regime, it is the dominant strategy for the public firm to set price at T=1. Public firm wants to be the leader in price because the greater welfare is obtained than otherwise. However, the private firm's strategy differs between discriminatory input pricing and uniform pricing. As stated in Lemma 1 (2), the private firm faces the highest cost handicap against the public firm when it is the price leader and the lowest when it is the follower, so that $\pi_1^{Pub^*} > \pi_1^{S^*} > \pi_1^{Pri^*}$. Instead of setting the price at T=1, the private firm will choose T=2 as the dominant strategy, reducing the cost handicap imposed by the upstream supplier and consequently achieving higher profits than otherwise. This means that the private firm earns higher profits as the follower than it would as the leader or simultaneous mover.

Proposition 3 states that standard conclusions about endogenous timing in a mixed oligopoly altered if discriminatory input pricing is adopted. In addition, the public leadership, as equilibrium outcome of endogenous timing game, does not ensure the highest consumer surplus

than other cases if goods are close substitutes, that is, $CS^{Pri^*} > CS^{Pub^*} > CS^{S^*}$ if $b \in \left(0.641, \frac{2}{3}\right)$.

In establishing competition policy, it is crucial for the government to clarify the market structure. Our model suggests policy implication on whether to ban the price discrimination of the foreign upstream manufacturer or not. By comparing equilibrium payoffs in the uniform input pricing and discriminatory input pricing, we can obtain the following proposition on government policy option.

Proposition 4: Suppose firms compete in prices (Bertrand) in downstream mixed oligopoly. In this case, banning price discrimination on imported inputs is desirable in terms of both social welfare and profits of the private firm.

Proof: A simultaneous-move, where equilibrium payoffs are $(W^{S^+}, \pi_1^{S^+})$, occurs under uniform input pricing (Lemma 5), while public leadership, where equilibrium payoffs are $(W^{Pub^*}, \pi_1^{Pub^*})$, occurs under discriminatory input pricing (Proposition 3). From Table 1, we have

$$\begin{split} & W^{S^{*}} - W^{P_{ub}^{*}} \\ &= \frac{b(1-b)\Gamma^{2}(512 - 448b - 1024b^{2} + 1143b^{3} + 561b^{4} - 1094b^{5} + 102b^{6} + 465b^{7} - 175b^{8} - 74b^{9} + 40b^{10})}{8(2-b^{2})^{2}(32 - 41b^{2} + 13b^{4})^{2}} > 0, \\ & \pi_{1}^{S^{*}} - \pi_{1}^{P_{ub}^{*}} \\ &= \frac{b(1-b)\Gamma^{2}(8 + 7b - 15b^{2} - 4b^{3} + 6b^{4})(64 - 8b - 97b^{2} + 8b^{3} + 45b^{4} - 2b^{5} - 6b^{6})}{4(2-b^{2})^{2}(32 - 41b^{2} + 13b^{4})^{2}} > 0. \end{split}$$

2. CASE 2: $b \in \left[\frac{2}{3}, 1\right)$

As mentioned in Footnote 3, if the private firm is Stackelberg leader in price and $b \in \left[\frac{2}{3}, 1\right)$, then only the public firm serves the market $(x_1^{p_{fi}*}=0)$. This can be explained as follows. Basically, private firms maximize profits and want to reduce market competition in an oligopoly market. When the private firm is the price leader, it will set a higher price and produce a lower output than under simultaneous case. In addition, the magnitude of cost handicap of the private firm is the largest when the private firm is the price leader. Compared to simultaneous-move case, this induces a production shift from the private to the public firm. This production shift would increase as the products become closer substitutes (i.e., an increase in *b*). In our model, a sufficiently large substitutability $\left(b \ge \frac{2}{3}\right)$ will lead to zero production by the private firm when the private firm is price leader $(x_1^{p_{fi}*}=0)$. The following proposition is obtained.

Proposition 5: If goods are close substitutes $\left(b \ge \frac{2}{3}\right)$ and the private firm is the leader in price, then the following equilibrium is obtained as a corner solution:

$$p_{0}^{Pri^{*}} = \frac{a+c}{2}, p_{1}^{Pri^{*}} = \frac{2a-b\Gamma}{2}, w_{0}^{Pri^{*}} = \frac{\Gamma}{2}, w_{1}^{Pri^{*}} = \frac{(2-b)\Gamma}{2},$$
$$x_{0}^{Pri^{*}} = \frac{a+c}{2}, \pi_{0}^{Pri^{*}} = \pi_{1}^{Pri^{*}} = 0, CS^{Pri^{*}} = W^{Pri^{*}} = \frac{\Gamma^{2}}{8}.$$

Proof: Solving $x_1^{Pri}(\mathbf{w}) = 0$ for w_1 in Eq. (9.2) yields $w_1 = bw_0 + (1-b)\Gamma$. By substituting this into $x_0^{Pri}(\mathbf{w})$ in Eq. (9.2), we get $x_0^{Pri}(w_0) = \Gamma - w_0$. And the maximization problem of upstream manufacturer is $\operatorname{Max}_{w_0} w_0 x_0^{Pri}(w_0)$, which yields $w_0^{Pri^*} = \frac{\Gamma}{2}$ and $w_1^{Pri^*} = \frac{(2-b)\Gamma}{2}$. Once $w_0^{Pri^*}$ and

75

 $w_1^{Pri^*}$ are obtained, $p_0^{Pri^*} = \frac{a+c}{2}$, $p_1^{Pri^*} = \frac{2a-b\Gamma}{2}$, and $x_0^{Pri^*} = \frac{a+c}{2}$ are given by Eqs. (9.1) and (9.2). Since $\pi_0^{Pri^*} = bx_0^{Pri^*} x_1^{Pri^*}$ and $\pi_1^{Pri^*} = (x_1^{Pri^*})^2$, $x_1^{Pri^*} = 0$ implies $\pi_0^{Pri^*} = \pi_1^{Pri^*} = 0$. Consumer surplus is $CS^{Pri^*} = \frac{\Gamma^2}{8}$, which equals to social welfare because producer surplus is zero (i.e., $\sum_{i=0,1} \pi_i^{Pri^*} = 0$).

Then, what will be the firms' timing of moves at equilibrium if goods are close substitutes? In this regard, we can see that the rankings on social welfare in Proposition 2(2) and that on profits of the private firm in Proposition 1(4) still hold in $b \in \left[\frac{2}{3}, 1\right)$, implying that public leadership is SPNE in observable delay game even when $b \in \left[\frac{2}{3}, 1\right)$.

Lemma 6: Suppose that goods are close substitutes $\left(b \ge \frac{2}{3}\right)$ and firms compete in price. In this case, if discriminatory input pricing is adopted, "public leadership" is the Nash equilibrium in observable delay game, suggesting that Proposition 3 holds irrespective of the value of *b*.

VI. The Case of Cournot Competition

In this section, we examine the case of downstream Cournot competition. In the simultaneous-move case, the profit maximization problem of firm 1 is $Max_{x_1}\pi_1(\mathbf{x}; w_1)$, while that of firm 0 is $Max_{x_0}W(\mathbf{x}; \mathbf{w})$. Solving the respective first order condition yields $\phi_0(x_1; w_0) = a - bx_1 - (c + w_0)$ and $\phi_1(x_0; w_1) = \frac{1}{2}[a - bx_0 - (c + w_1)]$, where ϕ_i , i = 0, 1, is firm *i*'s reaction function.¹³ Given $\mathbf{w} = (w_0, w_1)$, the equilibrium output under each leader-follower relationship is given by:

$$x_{0}^{s}(\mathbf{w}) = \frac{2\{a - (c + w_{0})\} - b\{a - (c + w_{1})\}}{2 - b^{2}}, x_{1}^{s}(\mathbf{w}) = \frac{\{a - (c + w_{1})\} - b\{a - (c + w_{0})\}}{2 - b^{2}} \quad (11.1)$$

$$x_{0}^{Pri}(\mathbf{w}) = \frac{(2 - b^{2})\{a - (c + w_{0})\} - b\{a - (c + w_{1})\}}{2(1 - b^{2})}, x_{1}^{Pri}(\mathbf{w}) = \frac{\{a - (c + w_{1})\} - b\{a - (c + w_{0})\}}{2(1 - b^{2})}. \quad (11.2)$$

$$x_{0}^{Pub}(\mathbf{w}) = \frac{4\{a - (c + w_{0})\} - 3b\{a - (c + w_{1})\}}{4 - 3b^{2}}, x_{1}^{Pub}(\mathbf{w}) = \frac{2[\{a - (c + w_{1})\} - b\{a - (c + w_{0})\}]}{4 - 3b^{2}}, \quad (11.3)$$

First, we examine the case where uniform input pricing is adopted. Solving the maximization problem, $\operatorname{Max}_{w}\sum_{i=0,1} x_{i}^{I}(w, w)w$, where $x_{i}^{I}(w, w)$ is given in Eqs. (11.1)-(11.3) with $w_{0}=w_{1}=w$, gives $w^{s+}=w^{Pub+}=\frac{\Gamma}{2}$. The equilibrium input price remains unchanged

2020]

¹³ Note that marginal cost pricing is confirmed from the reaction function of the public firm (i.e., $p_0 = c + w_0$).

irrespective of firms' role in the leader-follower relationship. Using input prices, we obtain the equilibrium market values under downstream Cournot competition as follows:

$$x_{0}^{s+} = \frac{(2-b)\Gamma}{2(2-b^{2})}, x_{1}^{s+} = \frac{(1-b)\Gamma}{2(2-b^{2})}, \pi_{1}^{s+} = \frac{(1-b)^{2}\Gamma^{2}}{4(2-b^{2})^{2}}, W^{s+} = \frac{(7-6b-2b^{2}+2b^{3})\Gamma^{2}}{8(2-b^{2})^{2}}, (12.1)$$

$$x_{0}^{Pri+} = \frac{(2+b)\Gamma}{4(1+b)}, x_{1}^{Pri+} = \frac{\Gamma}{4(1+b)}, \pi_{1}^{Pri+} = \frac{(1-b)\Gamma^{2}}{16(1+b)}, W^{Pri+} = \frac{(7+b)\Gamma^{2}}{32(1+b)},$$
(12.2)

$$x_{0}^{Pub+} = \frac{(4-3b)\Gamma}{2(4-3b^{2})}, x_{1}^{Pub+} = \frac{(1-b)\Gamma}{4-3b^{2}}, \pi_{1}^{Pub+} = \frac{(1-b)^{2}\Gamma^{2}}{(4-3b^{2})^{2}}, W^{Pub+} = \frac{(7-6b)\Gamma^{2}}{8(4-3b^{2})},$$
(12.3)

We are now ready to examine the first stage of the game — the endogenous order of moves in a mixed oligopoly. From Eqs. (12.1)-(12.3), the following lemma is obvious.

Lemma 7: Suppose that the downstream firms compete in quantity in a mixed duopoly. If the upstream monopolist adopts uniform input pricing, then both "public leadership" and "private leadership" are Nash equilibria.

Proof:
$$W^{Pri+} - W^{S+} = \frac{b^2(1-b)(4-b^2)\Gamma^2}{32(1+b)(2-b^2)^2} > 0$$
 and $W^{Pub+} - W^{S+} = \frac{b^2(1-b)^2\Gamma^2}{8(2-b^2)^2(4-3b^2)} > 0$ for firm

0 and
$$\pi_1^{Pri+} - \pi_1^{S+} = \frac{b^4(1-b)\Gamma^2}{16(1+b)(2-b^2)^2} > 0$$
 and $\pi_1^{Pub+} - \pi_1^{S+} = \frac{b^2(1-b)^2(8-5b^2)\Gamma^2}{4(8-10b^2+3b^4)^2} > 0$ for firm 1

are obtained from Eqs. (12.1)-(12.3). This implies that neither firm wants to determine its output in the same period as the rival firm. Therefore, both "public leadership" and "private leadership" are SPNE.

Lemma 7, which corresponds to Lemma 5 for Bertrand competition, confirms the results of Pal (1998). The intuition for Lemma 7 is as follows. Suppose all firms produce in period 1. In this case, if the public firm chooses period 2 instead and acts as the follower, then the output of the private firm increases $(x_1^{Pri+} > x_1^{S+})$ but that of the public firm decreases $(x_0^{Pri+} < x_0^{S+})$. This leads to an increase in total production $\left(\sum_i x_i^{Pri+} > \sum_i x_i^{S+}\right)$ so that social welfare is greater $(W^{Pri+} > W^{S+})$. Since the output of the private firm increases, its profit also increases $(\pi_1^{Pri+} > \pi_1^{S+})$. Alternatively, suppose that all firms produce in period 2. In this case, if the public firm moves into period 1 and acts as the leader, it would produce less output $(x_0^{Pub+} < x_0^{S+})$ and earn more profit $(\pi_0^{Pub+} > \pi_0^{S+})$ than it does in the simultaneous-move game. Since, in the Cournot model, quantities are strategic substitutes, this will lead to an increase in the private firm's profit $(\pi_1^{Pub+} > \pi_1^{S+})$ through output increase. An increase in producer surplus $\left(\sum_i \pi_i^{Pri+} > \sum_i \pi_i^{S+}\right)$ dominates the consumer surplus loss, resulting in an increase in social welfare $(W^{Pub+} > W^{S+})$.

Next, we turn to the case of discriminatory input pricing. Solving the maximization problem of the upstream manufacturer, $\max_{w_0, w_1} \sum_{i=0,1} w_i x_i^{I}(\mathbf{w})$, we obtain

$$W_i^{S^*} = W_i^{Pri^*} = \frac{a-c}{2}, i=0, 1,$$
 (13.1)

$$w_1^{Pub^*} = \frac{(a-c)(16+4b-15b^2)}{32-25b^2} > \frac{a-c}{2} > \frac{2(a-c)(8-5b-5b^2)}{32-25b^2} = w_0^{Pub^*}$$
(13.2)

As in Eq. (13.2), upstream manufacturer handicaps the private firm and subsidize the public firm using input pricing only when the public firm is the leader in the downstream Cournot competition. Since input prices for simultaneous-move and private leadership game are the same as those in uniform pricing, market outcomes for both cases under discriminatory input pricing are the same as those under uniform input pricing. The input prices under discriminatory pricing differ from those under uniform input pricing only when the market is under public leadership. By substituting Eq. (13.2) into market variables, we obtain equilibrium values for those variables under public leadership, as summarized in Table 3.¹⁴

Proposition 6: Suppose that downstream firms compete in quantity under discriminatory input pricing. Then, (1) both $\pi_1^{Pri^*} > \pi_1^{S^*} > \pi_1^{Pub^*}$ and $W^{Pri^*} > W^{S^*} > W^{Pub^*}$ holds. (2) In the observed delay game in Table 1, "private leadership" is the subgame-perfect Nash equilibrium (SPNE) in the mixed duopoly.

Proof: See Appendix 5.

Considering Eqs. (13.1) and (13.2), $\pi_1^{Pri^*} > \pi_1^{S^*} > \pi_1^{Pub^*}$ in Proposition 6(1) is straightforward. Since $w_i^{S^*} = w_i^{Pri^*} = \frac{a-c}{2}$ for $i=0, 1, \pi_1^{S^*} < \pi_1^{Pri^*}$ holds as in uniform pricing. However, since the private firm has a cost handicap against the public firm due to the upstream supplier's input pricing (Eq. (13.2)), the private firm's profits is lower in the public leadership compared to simultaneous-moves (i.e., $\pi_1^{Pub^*} < \pi_1^{S^*}$). As to social welfare, $W^{S^*} < W^{Pri^*}$ holds as in uniform pricing because $w_i^{S^*} = w_i^{Pri^*} = \frac{a-c}{2}$. However, the public firm, as a Stackelberg leader, produces less output with public leadership than in the simultaneous case, resulting in a relative increase in the private firm's output. This tends to raise the price level of the public firm. Therefore, the overall price level under public leadership rises above that with simultaneous-move case, resulting in $CS^{Pub^*} < CS^{S^*}$. Social welfare, reflecting consumer surplus, is lower in public leadership compared to the simultaneous-move case, $W^{Pub^*} < W^{S^*}$.

The background intuition of Proposition 6(2) is as follows. In the uniform input pricing regime, both firms' strategy is to produce at different time periods than their rival chooses. That is, either public leadership or private leadership can be a Nash equilibrium (Lemma 7). We show that public leadership cannot be a SPNE under discriminatory input pricing in a sequential game. Note that the private firm faces a handicap against the public firm in terms of input price only when the market is under public leadership (Eq. (13.2)). If the market is under public leadership, the private firm, by moving to period 1, can remove the cost handicap in input prices and increase its profits (i.e., $\pi_1^{S*} > \pi_1^{Pub^*}$). Similarly, public leadership is not attractive to the public firm either. The public firm has an incentive to move to period 2 (i.e.,

2020]

¹⁴ Just as with price competition in the downstream sector, when downstream firms compete in quantity, a corner solution emerges. In the sub-game of a public leadership, if $b \in \left[\frac{4}{5}, 1\right)$, $x_0^{Pub^*} > 0$ and $x_1^{Pub^*} = 0$ are derived as a corner solution. That is, only the public firm will serve the market in equilibrium.

	TABLE 3. MARKET EQU	ILIBRIUMS UNDER EACH RE	GIME: DOWNSTREAM QUANTITY (COMPETITION
			Public Leadership	
	Simultaneous Move	Private Leadership	$b \in \left(0, \frac{4}{5}\right)$	$b \in \left[\frac{4}{5}, 1\right)$
Input price	$w_0^{S^*} = w_1^{S^*} = \frac{\Gamma}{2}$	$w_0^{P,n} = w_1^{P,n} = \frac{\Gamma}{2}$	$w_{1^{bh*}}^{b_{hh*}} = \frac{2(8-b-5b^2)\Gamma}{32-25b^2}$ $w_{1^{bh*}}^{p_{hh*}} = \frac{(16+4b-15b^2)\Gamma}{32-25b^2}$	$w_{1}^{p,ms} = \frac{\Gamma}{2}$ $w_{1}^{p,ms} = \frac{(2-b)\Gamma}{2}$
Output price	$p_{i}^{s} = \frac{a+c}{2}$ $p_{i}^{s} = \frac{a(3-b-b^{2})+c(1+b-b^{2})}{2(2-b^{2})}$	$p_0^{p_0*} = \frac{a+c}{2}$ $p_1^{p_0*} = \frac{a(3-b)+c(1+b)}{4}$	$p_{p_{\text{tot}}^{p_{\text{tot}}}}^{p_{\text{tot}}} = \frac{a(16+2b-15b^2)+2c(8-b-5b^2)}{32-25b^2}$ $p_{p_{\text{tot}}}^{p_{\text{tot}}} = \frac{3a(8-2b-5b^2)+2c(4+3b-5b^2)}{32-25b^2}$	$p_{0,m^*}^{n_{0,m^*}} = \frac{a+c}{2}$ $p_{1,m^*}^{n_{0,m^*}} = \frac{a(2-b)+bc}{2}$
Output	$x_1^{5^*} = \frac{(2-b)\Gamma}{2(2-b^*)}$ $x_1^{5^*} = \frac{(1-b)\Gamma}{2(2-b^*)}$	$x_0^{p,n} = \frac{(2+b)\Gamma}{4(1+b)}$ $x_1^{p,n} = \frac{\Gamma}{4(1+b)}$	$x_{0}^{p_{ab}*} = \frac{2(8-5b)\Gamma}{32-25b^2}$ $x_{1}^{p_{ab}*} = \frac{2(4-5b)\Gamma}{32-25b^2}$	$x_0^{Dut} = \frac{\Gamma}{2}$ $x_0^{Dut} = 0$
Profit	$\pi_1^{s} = 0$ $\pi_1^{s} = (x_1^{s})^2 = \frac{(1-b)^2 \Gamma^2}{4(2-b^2)^2}$	$\begin{aligned} \pi_0^{p,r} = 0 \\ \pi_1^{p,r} = (1-b^2)(x_1^{p,r})^2 = \frac{(1-b)\Gamma^2}{16(1+b)} \end{aligned}$	$\pi_{0}^{P,m^{*}} = \frac{b_{2} x_{0}^{P,m^{*}} x_{1}^{P,m^{*}}}{2} \frac{2b(8-5b)(4-5b)(1-5b)^{2}}{(32-25b^{2})^{2}}$ $\pi_{1}^{P,m^{*}} = (x_{1}^{P,m^{*}})^{2} = \frac{4(4-5b)^{2} \Gamma^{2}}{(32-25b^{2})^{2}}$	$\pi_1^{p_{abs}}=0$ $\pi_1^{p_{abs}}=0$
welfare	$CS^{sy} = \frac{(5 - 2b - 4b^2 + 2b^3)\Gamma^2}{8(2 - b^2)^2}$ $W^{sy} = \frac{(7 - 6b - 2b^2 + 2b^3)\Gamma^2}{8(2 - b^2)^2}$	$CS^{p_{tris}} = \frac{(5+3b)\Gamma^2}{32(1+b)}$ $W^{p_{tris}} = \frac{(7+b)\Gamma^2}{32(1+b)}$	$CS^{P_{M}s} = \frac{4(40-28b-35b^2+25b^3)\Gamma^2}{(32-25b^2)^2}$ $W^{P_{M}s} = \frac{(32-25b^2)^2}{(112-104b-80b^2+75b^3)\Gamma^2}$ $(32-25b^2)^2$	$CS^{habs} = \frac{\Gamma^2}{8}$ $W^{habs} = \frac{\Gamma^2}{8}$
<i>Note</i> : $\Gamma \equiv a$	<i>2</i> –			

ŭ õ Ĕ R μ Ц Ž 3

[June

the simultaneous-move case) under public leadership. Although the public firm loses its "leadership" position, it can raise social welfare $(W^{S^*} > W^{Pub^*})$. On the other hand, when the market is under private leadership, no firm has an incentive to deviate from its strategy.

The above Proposition 6(2) is noteworthy from the following viewpoint. The indeterminacy of the firms' role with respect to the leader-follower relationship vanishes in a two-tier mixed oligopoly, in sharp contrast with Pal (1998) who demonstrated multiple equilibria in sequential games when both firms compete on quantity in a one-tier mixed oligopoly.

Should the government ban discriminatory input pricing by the foreign upstream monopolist? We offer the following proposition on government policy with respect to price discrimination.

Proposition 7: Suppose that firms compete in quantity (Cournot) in a downstream mixed oligopoly. In this case, allowing price discrimination on imported inputs is socially desirable.

Proof: We show that the equilibrium social welfare under discriminatory input pricing (W^{Pri^*}) is greater than or equal to the social welfare under uniform input pricing $(W^{Pri^+}$ or $W^{Pub^+})$. Since $w_i^{Pri^*} = w^+ = \frac{a-c}{2}$ in Eq. (13.1), $W^{Pri^*} = W^{Pri^+}$ holds. In addition, $W^{Pri^*} - W^{Pub^+} = W^{Pri^+} - W^{Pub^+} > 0$ because $W^{Pri^+} - W^{Pub^+} = \frac{3b^2(1-b)\Gamma^2}{32(1+b)(4-3b^2)} > 0$ from Eqs. (12.2) and (12.3).

Proposition 7, which corresponds to Proposition 4 for the Bertrand competition, suggests that, in a quantity-setting mixed oligopoly, it is socially desirable for the domestic government not to intervene in the input pricing behavior (i.e., allow discriminatory input pricing) by the foreign upstream monopolist than enforcing uniform input pricing. This is because, if price discrimination was banned, then there is a possibility that firms' strategic behavior in endogenous timing game may result in socially undesirable one (that is, public firm leadership) of the two equilibria. Allowing discriminatory input pricing of the foreign upstream monopolist has the effect of blocking the possibility that the public firm leadership, which is socially less efficient than the private leadership, is realized as an equilibrium market structure.

VII. Extension: Domestic Upstream Supplier

To see how the upstream firm's nationality affects the main results, we assume in this section that the upstream supplier is a domestic firm. In this case, the social welfare is $SW = \frac{1}{2}\{(1-\gamma)(x_0^2+x_1^2)+(x_0+x_1)^2\}+\sum_{i=0,1}(p_i-c)x_i$, where the producer surplus is represented as the sum of the aggregated profits of vertically related firms with marginal cost c. First, we consider price competition. Given $\mathbf{w} = (w_0, w_1)$, the equilibrium output in the last stage of the game under each leader-follower relationship is given by

$$x_{0}^{s} = x_{0}^{Pri} = \frac{a-c}{1+b}, x_{0}^{Pub}(w_{1}) = \frac{(4+b-3b^{2}-b^{3})(a-c)+b(1+b)w_{1}}{(1+b)(4-3b^{2})}$$
(14.1)
$$x_{1}^{s}(w_{1}) = \frac{a-c-(1+b)w_{1}}{(1+b)(2-b^{2})}, x_{1}^{Pri}(w_{1}) = \frac{(a-c)-(1+b)w_{1}}{2(1+b)},$$

$$x_1^{Pub}(w_1) = \frac{(2-b^2)(a-c) - 2(1+b)w_1}{(1+b)(4-3b^2)}.$$
(14.2)

In the equations, the following points are noteworthy. First, the public firm produces a fixed amount of output irrespective of input prices $\left(x_0^S = x_0^{Pri} = \frac{a-c}{1+b}\right)$ with both simultaneous-moves and private-firm leadership.¹⁵ Second, each firm's equilibrium output depends only on the input price charged to the private firm (w_1) .

In the second stage of the game, the upstream supplier charges the private firm the monopoly price for the input $\left(w_1^{S^*}=w_1^{Pri^*}=\frac{(a-c)}{2(1+b)}\right)$ when the market structure is based on either simultaneous-moves or private leadership. However, when market is under public leadership, the upstream supplier imposes the upper bound input price to the private firm¹⁶ (i.e., $w_1^{Pub^*}=\frac{(2-b^2)(a-c)}{2(1+b)}$), reducing the private firm's output to zero, so that only the public firm remains in the market. However, the possibility of private-firm production — even if no actual production takes place — forces the public firm to choose a larger output. Once equilibrium input prices are established, the equilibrium market values are obtained. As a result, when downstream firms engage in Bertrand competition, the social welfare and profit rankings for different leader-follower relationships are obtained as follows:

$$\pi_{1}^{Pri^{*}} = \frac{\Gamma^{2}}{16(1+b)^{2}} > \frac{(1-b)\Gamma^{2}}{4(1+b)(2-b^{2})^{2}} = \pi_{1}^{S^{*}} > \pi_{1}^{Pub^{*}} (=0), \qquad (15.1)$$

$$W^{S^{*}} = \frac{(23+32b-20b^{2}-32b^{3}+4b^{4}+8b^{5})\Gamma^{2}}{8(1+b)^{2}(2-b^{2})^{2}} > \frac{(23+32b)\Gamma^{2}}{32(1+b)^{2}} = W^{Pri^{*}}, \qquad (15.2)$$

$$W^{S^{*}} > \frac{(4+8b+3b^{2})\Gamma^{2}}{8(1+b)^{2}} = W^{Pub^{*}}$$

In above equations, the profit ranking of the private firm shows the same result as in Barcena-Ruiz (2007), but the welfare ranking is different. Unlike the welfare ranking $(W^{Pub^*} > W^{S^*} > W^{Pri^*})$ in the one-tier mixed oligopoly analyzed by Barcena-Ruiz (2007), the welfare under public leadership is less than that in the simultaneous-move case. When the market is under public leadership, the negative effects of market distortion on social welfare caused by input price discrimination are large enough to reduce the welfare level below the simultaneous-move case.

We now consider Cournot competition. In this case, nonnegative profit constraint for the

81

[June

¹⁵ If firm 0 behaves optimally, the reaction function $p_0[p_1] = c + b(p_1 - c)$ is derived from $\frac{\partial SW}{\partial p_0} = 0$, and substituting this into Eq. (2) yields the equilibrium output of firm 0, which is independent of input prices.

¹⁶ Since $\frac{\partial q_0^{Pub}}{\partial w_1} > 0$, it is advantageous for the upstream manufacturer to offer the private firm as high an input price as possible, thereby shifting all the output production from the private to the public firm, whose derived demand for intermediate input is inelastic.

public firm is required.¹⁷ The nonnegative profit constraint is binding for all leader-follower relationships (i.e., $\pi_0^{S^+} = \pi_0^{Pub^+} = \pi_0^{Pri^+} = 0$). Given $\mathbf{w} = (w_0, w_1)$, the equilibrium output in the last stage game is

$$x_{0}^{S}(\mathbf{w}) = x_{0}^{Pub}(\mathbf{w}) = \frac{2\{a - (c + w_{0})\} - b\{a - (c + w_{1})\}}{2 - b^{2}},$$

$$x_{0}^{Pri}(\mathbf{w}) = \frac{(2 - b^{2})\{a - (c + w_{0})\} - b\{a - (c + w_{1})\}}{2(1 - b^{2})}$$
(16.1)
$$\{a - (c + w_{1})\} - b\{a - (c + w_{0})\} - b\{a - (c + w_{0})\} - b\{a - (c + w_{0})\} - b\{a - (c + w_{0})\}$$

$$x_{1}^{s}(\mathbf{w}) = x_{1}^{Pub}(\mathbf{w}) = \frac{\{a - (c + w_{1})\} - b\{a - (c + w_{0})\}}{2 - b^{2}}, x_{1}^{Pri}(\mathbf{w}) = \frac{\{a - (c + w_{1})\} - b\{a - (c + w_{0})\}}{2(1 - b^{2})}.$$
(16.2)

From the equations, the equilibrium output in the public leadership is the same as that in the simultaneous-move case (i.e., $x_i^{Pub} = x_i^s$, i=0, 1). Since non-negative profit constraint is binding, the public firm cannot make use of its strategic advantage of moving first, so that the market equilibrium is the same for public-firm leadership and the simultaneous case. In the second stage game, solving the maximization problem $\left(i.e., \max_{w_0, w_1} \sum_{i=0, 1} w_i x_i^i(\mathbf{w})\right)$ gives $w_i^{S^*} = w_i^{Pub^*} = w_i^{Pri^*} = \frac{a-c}{2}$ for i=0, 1. Using equilibrium input prices, we obtain the following relationships for social welfare and firm 1's profits when downstream firms compete in quantity:

$$\pi_1^{Pri^*} = \frac{(1-b)\Gamma^2}{16(1+b)} > \frac{(1-b)^2\Gamma^2}{4(2-b^2)^2} = \pi_1^{S^*} = \pi_1^{Pub^*},$$
(17.1)

$$W^{Pri^{*}} = \frac{(19+5b)\Gamma^{2}}{32(1+b)} > \frac{(19-14b-8b^{2}+6b^{3})\Gamma^{2}}{8(2-b^{2})^{2}} = W^{S^{*}} = W^{Pub^{*}}.$$
(17.2)

The private firm increases its output (as compared to the simultaneous case) when it is the leader because it knows that the follower (the public firm) will reduce its output, which means that $q_1^{Pri^*} > q_1^{S^*} = q_1^{Pub^*}$ and $\pi_1^{Pri^*} > \pi_1^{S^*} = \pi_1^{Pub^*}$. As to welfare ranking, $W^{S^*} = W^{Pub^*}$ holds because $q_i^{S^*} = q_i^{Pub^*}$ and $w_i^{S^*} = w_i^{Pub^*}$ for i = 0, 1. $W^{Pri^*} > W^{S^*} (= W^{Pub^*})$ is explained as follows. Suppose that both firms produce in period 1 (i.e., T = 1). In this case, if the public firm moves to T = 2, then the output and profits of the private firm increase while those of the public firm remain unchanged (as compared to the simultaneous case), so that the producer surplus increases $(PS^{Pri^*} > PS^{S^*})$. Because total output increases, the consumer surplus tends to rise above the simultaneous-move level $(CS^{Pri^*} > CS^{S^*})$. As a result, $W^{Pri^*} > W^{S^*} (= W^{Pub^*})$.

In view of Eqs. (15.1), (15.2), (17.1) and, (17.2), the following proposition is offered.

Proposition 8: Suppose that the upstream manufacturer is a domestic firm. When downstream

2020]

¹⁷ If downstream firms behave optimally without constraints, the public firm produces up to the level where the price equals the marginal cost of the upstream manufacturer (i.e., $p_0 = c$), resulting in negative profits for the public firm. In addition, the private firm is charged extremely high input prices by the upstream manufacturer and, thus, is barred from production. As a result, when both firms behave optimally without any constraint à la Cournot in the market, only the public firm remains in the market for all leader-follower relationships. It is useless to discuss mixed oligopoly in such a circumstance.

firms compete a là Bertrand (resp. Cournot) in an observable delay game, a simultaneous-move (resp. sequential-move) game is the SPNE in a mixed duopoly.

Proof: It is obvious from Eqs. (15.1) and (15.2) for Bertrand competition and from Eqs. (17.1) and (17.2) for Cournot competition.

The above proposition confirms the results of Pal (1998) and Barcena-Ruiz (2007) for Cournot and Bertrand competition, respectively. Therefore, if the upstream manufacturer is a domestic firm, the endogenous order of firms' moves is consistent between a two-tier mixed oligopoly with upstream monopolist and a standard one-tier mixed oligopoly, as long as nonnegative profit constraint is introduced when downstream market is under Cournot competition.

VIII. Concluding Remarks

Our main results are as follows. First, we have shown that, under discriminatory input pricing, the upstream supplier handicaps the private firm but subsidizes the public firm through input pricing. The cost handicap of the private firm is largest when the private (resp. public) firm is the leader in the downstream Bertrand (resp. Cournot) market. In addition, this cost handicap affects firms' move-timing decisions in an observable delay game, leading to public (resp. private) leadership in downstream Bertrand (resp. Cournot) competition. This result sharply contrasts with the simultaneous-move equilibrium in a price-setting (Bárcena-Ruiz, 2007) and sequential-move multiple equilibria in a quantity-setting (Pal, 1998) mixed oligopoly described in previous research. Second, when the downstream market is characterized by Bertrand competition, the rankings for firm's profit and consumer surplus are reversed but welfare ranking remains unchanged between uniform and discriminatory input pricing. Somewhat surprisingly, under discriminatory input pricing, the private firm's profits are highest (resp. lowest), in contrast to the ranking under uniform pricing, when the private firm is the follower (resp. leader) in the market. Third, the nationality of the upstream supplier makes a difference for the firms in terms of timing their moves. With a domestic upstream manufacturer, the endogenous order of firms' moves is consistent between a two-tier mixed oligopoly with upstream monopolist and a standard one-tier mixed oligopoly, as long as nonnegative profit constraint is introduced when downstream market is under Cournot competition. This implies that as long as the nonnegative profit constraint is satisfied, the endogenously determined order of firms' moves does not substantially change if the upstream monopolist in a vertically related market is a domestic firm. Fourth, our analysis has implications for competition policy. According to our analysis, policy makers need to consider the competition mode (i.e., Cournot or Bertrand) in the downstream market before making any decision on whether to ban input price discrimination by the foreign upstream manufacturer. That is, banning price discrimination for imported inputs is desirable, from the viewpoint of both social welfare and firms' profit, with downstream Bertrand competition, but input price discrimination is socially desirable with downstream Cournot competition.

The conclusions of our paper depends largely on critical assumptions to keep the model as simple and transparent as possible, such as linear demand function, constant marginal cost, full bargaining power of upstream manufacturer, simple vertical structure that upstream manu-

83

facturer belongs to foreign country. Especially noteworthy is that the public firm in our model is assumed as a simple welfare maximizer, and hence, we could not discuss the privatization problem that has been one of key issues in mixed oligopoly literature. One way to incorporate the privatization issue into the study is to introduce partial privatization approach formulated by Matsumura (1998) and analyze the optimal degree of privatization, which will provide a richer policy implications, particularly with respect to the interdependence between the public policy toward price discrimination in the upstream market and privatization policy in the downstream market. Our future follow-up to this study will focus on solving the above-mentioned problems.

Appendix

1. Proof of Lemma 2

By substituting $w^{I+} = \frac{a-c}{2}(I=S,Pub,Pri)$ into Eqs. (5.1) and (5.2) for the simultaneous-move game, into Eqs. (7.1) and (7.2) for the public leadership, and into Eqs. (9.1) and (9.2) for the private leadership, we obtain the equilibrium outcomes in the three subgames under uniform input pricing as follows:

$$\begin{split} p_{0}^{5^{+}} &= \frac{2ab(1-b)+(2-b)(a+c)}{2(2-b^{2})}, p_{1}^{5^{+}} &= \frac{2a(1-b)+(1+b-b^{2})(a+c)}{2(2-b^{2})}, \\ \pi_{0}^{5^{+}} &= \frac{b(1-b)(a-c)^{2}}{4(1+b)(2-b^{2})}, \pi_{1}^{5^{+}} &= \frac{(1-b)(a-c)^{2}}{4(1+b)(2-b^{2})^{2}}, CS^{5^{+}} &= \frac{(5-b-3b^{2}+b^{3})(a-c)^{2}}{8(1+b)(2-b^{2})^{2}}, \\ W^{5^{+}} &= \frac{(7+b-7b^{2}-b^{3}+2b^{4})(a-c)^{2}}{8(1+b)(2-b^{2})^{2}}, p_{0}^{p_{0}b^{+}} &= \frac{a(4+b-4b^{2})+c(4-b-2b^{2})}{2(4-3b^{2})}, \\ p_{1}^{p_{ub}+} &= \frac{a(6-2b-4b^{2}+b^{3})+c(2+2b-2b^{2}-b^{3})}{2(4-3b^{2})}, \pi_{0}^{p_{ub}+} &= \frac{b(1-b)(4+b-3b^{2}-b^{3})(a-c)^{2}}{4(1+b)(4-3b^{2})^{2}}, \\ \pi_{1}^{p_{ub}+} &= \frac{(1-b)(2-b^{2})^{2}(a-c)^{2}}{4(1+b)(4-3b^{2})^{2}}, CS^{p_{ub}+} &= \frac{(5+b-3b^{2}-b^{3})(a-c)^{2}}{8(1+b)(4-3b^{2})}, \\ W^{p_{ub}+} &= \frac{(7+b-5b^{2}-b^{3})(a-c)^{2}}{4(1+b)(4-3b^{2})}, p_{0}^{p_{ri}+} &= \frac{a(2+3b)+c(2+b)}{4(1+b)}, p_{1}^{p_{ri}+} &= \frac{a(3+2b)+c(1+2b)}{4(1+b)}, \\ \pi_{0}^{p_{ri}+} &= \frac{b(a-c)^{2}}{8(1+b)^{2}}, \pi_{1}^{p_{ri}+} &= \frac{(a-c)^{2}}{16(1+b)^{2}}, CS^{p_{ri}+} &= \frac{(5+4b)(a-c)^{2}}{32(1+b)^{2}}, W^{p_{ri}+} &= \frac{(7+8b)(a-c)^{2}}{32(1+b)^{2}}, \end{split}$$

Lemma 2 can be easily verified by a simple calculation using the above equilibrium values.

2. Proof of Proposition 1

(1) From Table 1, we have:

$$p_{1}^{p_{f1}*} - p_{1}^{S^*} = \frac{b^2(8 - 4b - 8b^2 - b^3 + 3b^4)\Gamma}{(8 - 9b^2)(8 - 5b^2 + b^4)} > 0 \text{ and}$$

$$p_{1}^{S^*} - p_{1}^{p_{ub}*} = \frac{b(1 - b^2)^2(16 - 6b - 13b^2 + 3b^3 + 2b^4)\Gamma}{(8 - 5b^2 + b^4)(32 - 41b^2 + 13b^4)} > 0, \text{ which yields } p_{1}^{p_{f1}*} > p_{1}^{S^*} > p_{1}^{p_{ub}*}.$$
(2) From Table 1, we have:

$$x_{1}^{p_{ub}*} - x_{1}^{S^*} = \frac{b(1 - b)(16 - 6b - 7b^2 + 9b^3 - b^4 - 3b^5)\Gamma}{(8 - 5b^2 + b^4)(32 - 41b^2 + 13b^4)} > 0 \text{ and}$$

$$x_{1}^{s^{s}} - x_{1}^{p_{f}^{s}} = \frac{b^{2}(8 + 4b - 8b^{2} - b^{3} - 3b^{4})\Gamma}{(1 + b)(8 - 9b^{2})(8 - 5b^{2} + b^{4})} > 0, \text{ which yields } x_{1}^{p_{u}b^{*}} > x_{1}^{s^{*}} > x_{1}^{p_{f}^{s}} \text{ for } b \in \left(0, \frac{2}{3}\right).$$

Let $\Delta x^{l^{s}} \equiv x_{0}^{l^{s}} - x_{1}^{l^{s}}(I = Pri, Pub, S).$ From Table 1, we have $\Delta x^{Pub^{*}} = \frac{2(2 - b^{2})^{2}\Gamma}{(32 - 41b^{2} + 13b^{4})},$
 $\Delta x^{s^{*}} = \frac{2\Gamma}{(8 - 5b^{2} + b^{4})}, \text{ and } \Delta x^{Pri^{*}} = \frac{2\Gamma}{(8 - 9b^{2})}.$ Thus, $\Delta x^{Pri^{*}} - \Delta x^{s^{*}} = \frac{2b^{2}(4 + b^{2})\Gamma}{(8 - 9b^{2})(8 - 5b^{2} + b^{4})} > 0$
and $\Delta x^{s^{*}} - \Delta x^{Pub^{*}} = \frac{2b^{2}(11 - 19b^{2} + 9b^{4} - b^{6})\Gamma}{(8 - 5b^{2} + b^{4})(32 - 41b^{2} + 13b^{4})} > 0, \text{ implying } \Delta x^{Pri^{*}} > \Delta x^{s^{*}} > \Delta x^{Pub^{*}}.$

(3) From Table 1, aggregate output, $\sum_{i=0,1} x_i^{*}$, is obtained as follows:

$$\sum x_{i}^{Pub^{*}} = \frac{2(12+2b-14b^{2}-3b^{3}+4b^{4}+b^{5})\Gamma}{(1+b)(32-41b^{2}+13b^{4})}, \quad \sum x_{i}^{S^{*}} = \frac{2(3-b^{2})\Gamma}{(1+b)(8-5b^{2}+b^{4})}, \text{ and}$$

$$\sum x_{i}^{Put^{*}} = \frac{6(1-b)\Gamma}{(8-9b^{2})}. \text{ Thus,}$$

$$\sum x_{i}^{Pub^{*}} - \sum x_{i}^{S^{*}} = \frac{2b(1-b)^{3}(16+15b-4b^{2}-6b^{3}-b^{4})\Gamma}{(8-5b^{2}+b^{4})(32-41b^{2}+13b^{4})} > 0 \text{ and}$$

$$\sum x_{i}^{S^{*}} - \sum x_{i}^{Put^{*}} = \frac{2b^{2}(4-9b^{2}+3b^{4})\Gamma}{(1+b)(8-9b^{2})(8-5b^{2}+b^{4})} > 0, \text{ implying } \sum x_{i}^{Pub^{*}} > \sum x_{i}^{S^{*}} > \sum x_{i}^{Put^{*}}.$$

(4) We have $\pi_1^{Pub^*} - \pi_1^{S^*} = (1-b^2)(x_1^{Pub^*} + x_1^{S^*})(x_1^{Pub^*} - x_1^{S^*}) > 0$ because $x_1^{Pub^*} - x_1^{S^*} > 0$ from Proposition 1(2), and $\pi_1^{S^*} - \pi_1^{Prt^*} = \frac{b^3(256 - 288b - 384b^2 + 388b^3 + 100b^4 - 115b^5 + 17b^6 + 3b^7 - 9b^8)\Gamma^2}{(1+b)(8 - 9b^2)^2(8 - 5b^2 + b^4)^2} > 0$, implying $\pi_1^{Pub^*} > \pi_1^{S^*} > \pi_1^{Prt^*}$.

3. Proof of Lemma 4

From Table 1 we get the following relations:

$$p_0^{p_{7}*} - p_0^{5*} = \frac{b^3(4 - 10b + b^2 + 3b^3)\Gamma}{(8 - 9b^2)(8 - 5b^2 + b^4)} > [<]0 \text{ if } b \in (0, \ 0.447) \Big[b \in \Big(0.447, \frac{2}{3}\Big) \Big], \tag{A1}$$

$$p_0^{S^*} - p_0^{Pub^*} = \frac{b(1+b)(1-b)^2(16-12b-19b^2+6b^3+5b^4)\Gamma}{(8-5b^2+b^4)(32-41b^2+13b^4)} > 0,$$
(A2)

$$p_0^{p_{l}*} - p_0^{p_{l}b*} = \frac{b(16 - 28b - 15b^2 + 27b^3 + 4b^4 - 6b^5)\Gamma}{(8 - 9b^2)(32 - 41b^2 + 13b^4)} > [<]0 \text{ if } b \in (0, \ 0.582) \Big[b \in \Big(0.582, \frac{2}{3}\Big) \Big],$$
(A3)

From Eqs. (A1), (A2), and (A3), $p_0^{p_{rl}^*} \ge p_0^{S^*} \ge p_0^{p_{ub}^*}$ for $b \le 0.447$, $p_0^{S^*} \ge p_0^{p_{rl}^*} \ge p_0^{p_{ub}^*}$ for $0.447 \le b \le 0.582$, and $p_0^{S^*} \ge p_0^{p_{ub}^*} \ge p_0^{p_{rl}^*}$ for $0.582 \le b \le \frac{2}{3}$.

4. Proof of Proposition 2

(1) The following relationship is obtained with respect to consumer surplus from Table 1:

85

[June

86

$$CS^{Pub^*} - CS^{S^*} = \frac{b(1-b)^2(2-b^2) \binom{1536-320b-4032b^2+155b^3+3891b^4+315b^5}{-1713b^6-287b^7+345b^8+81b^9-27b^{10}-8b^{11}} \Gamma^2}{(8-5b^2+b^4)^2(32-41b^2+13b^4)^2} > 0,$$
(A4)

$$CS^{S^*} - CS^{Pri^*} = \frac{b^2 \left(\frac{128 - 624b^2 + 888b^4}{-488b^6 + 109b^8 - 9b^{10}} \right) \Gamma^2}{(1+b)(8-9b^2)^2 (8-5b^2 + b^4)^2} > [<]0 \text{ if } b \in (0, \ 0.587) \left[b \in \left(0.587, \ \frac{2}{3} \right) \right],$$
(A5)

$$CS^{Pub^{*}} - CS^{Pub^{*}} = \frac{b \begin{pmatrix} 3072 - 1664b - 15104b^{2} + 4982b^{3} \\ + 30256b^{4} - 5525b^{5} - 31552b^{6} \\ + 2735b^{7} + 18044b^{8} - 542b^{9} \\ - 5364b^{10} + 18b^{11} + 648b^{12} \end{pmatrix} \Gamma^{2}}{(1+b)(8-9b^{2})^{2}(32-41b^{2}+13b^{4})^{2}} > [<]0 \text{ if } b \in (0, \ 0.641) \Big[b \in \Big(0.641, \frac{2}{3}\Big) \Big].$$
(A6)

From Eqs. (A4) -(A6), it holds that $CS^{Pub^*} > CS^{S^*} > CS^{Prt^*}$ for b < 0.587, $CS^{Pub^*} > CS^{Prt^*} > CS^{S^*}$ for 0.587 < b < 0.641, and $CS^{Prt^*} > CS^{Pub^*} > CS^{S^*}$ for $0.641 < b < \frac{2}{3}$. (2) From Table 1, we obtain: $W^{Pub^*} - W^{S^*}$

$$=\frac{b(1+b)(1-b)^{2}(1024-640b-2432b^{2}+1778b^{3}+2416b^{4}-2000b^{5}-1298b^{6}+1123b^{7}+401b^{8}-310b^{9}-68b^{10}+33b^{11}+5b^{12})\Gamma^{2}}{(8-5b^{2}+b^{4})^{2}(32-41b^{2}+13b^{4})^{2}}>0,$$

and $W^{S*}-W^{Pri^{*}}=\frac{b^{2}(128-400b^{2}+524b^{4}-361b^{6}+131b^{8}-18b^{10})\Gamma^{2}}{(1+b)(8-9b^{2})^{2}(8-5b^{2}+b^{4})^{2}}>0,$ implying
 $W^{Pub^{*}}>W^{S^{*}}>W^{Pri^{*}}.$

5. Proof of Proposition 6

(1) In Eq. (13.1), $w_i^{p_{fl}*} = w_i^{S^*} = \frac{a-c}{2}$, implying that if market is either in private leadership or in simultaneous-move, then the market equilibria under discriminatory input pricing are the same as those under uniform input pricing. Therefore, $W^{p_{fl}*} = W^{P_{fl}+}$ and $W^{S^*} = W^{S^+}$ holds. From the proof of Lemma 7, we have $W^{p_{fl}+} > W^{S^+}$, implying $W^{p_{fl}*} > W^{S^*}$. Next, we show that $W^{S^*} > W^{p_{4b}*}$. From Table 3, we obtain

$$W^{S^*} - W^{Pub^*} = \begin{cases} \frac{b(512 - 960b + 192b^2 + 663b^3 - 486b^4 + 30b^5 + 50b^6)\Gamma^2}{8(64 - 82b^2 + 25b^4)^2} > 0 \text{ if } b \in \left(0, \frac{4}{5}\right) \\ \frac{(1 - b)^2(3 - b^2)\Gamma^2}{8(2 - b^2)^2} > 0 \text{ if } b \in \left(\frac{4}{5}, 1\right) \end{cases}$$

This leads to $W^{Pri^*} > W^{S^*} > W^{Pub^*}$ for the domain of $b \in (0, 1)$. (2) In Eq. (13.1), $w_i^{Pri^*} = w_i^{S^*} = \frac{a-c}{2}$, which leads to $\pi_1^{Pri^*} = \pi_1^{Pri^+}$ and $\pi_1^{S^*} = \pi_1^{S^+}$. In addition, we have $\pi_1^{Pri^+} > \pi_1^{S^+}$ in the proof of Lemma 7, which, in turn, implies $\pi_1^{Pri^*} > \pi_1^{S^*}$. Next, we will demonstrate

2020]

 $\pi_1^{S^*} > \pi_1^{Pub^*}$. From Table 3, we obtain:

$$\pi_{1}^{S^{*}} - \pi_{1}^{Pub^{*}} = \begin{cases} \frac{b(512 - 1152b + 640b^{2} + 369b^{3} - 610b^{4} + 225b^{5})\Gamma^{2}}{4(64 - 82b^{2} + 25b^{4})^{2}} > 0 \text{ if } b \in \left(0, \frac{4}{5}\right) \\ \pi_{1}^{S^{*}} > 0 \text{ if } b \in \left(\frac{4}{5}, 1\right) \end{cases}$$

Therefore, this leads to $\pi_1^{Pri^*} > \pi_1^{S^*} > \pi_1^{Pub^*}$ for the domain of $b \in (0, 1)$.

References

- Amin, M. and A. Islam (2015), "Use of Imported Inputs and the Cost of Importing: Evidence from Developing Countries," *Applied Economics Letters* 22, pp.488-492.
- Amir, R. and J.Y. Jin (2001), "Cournot and Bertrand Equilibria Compared: Substitutability, Complementarity and Concavity," *International Journal of Industrial Organization* 19, pp.303-317.
- Amir, R. and A. Stepanova (2006), "Second-Mover Advantage and Price Leadership in Bertrand Duopoly," *Games and Economic Behavior* 55, pp.1-20.
- Barcena-Ruiz, J.C. (2007), "Endogenous Timing in a Mixed Oligopoly: Price Competition," Journal of Economics 91, pp.263-272.
- Barcena-Ruiz, J.C. and M.B. Garzon (2010), "Endogenous Timing in a Mixed Oligopoly with Semipublic Firms," *Portuguese Economic Journal* 9, pp.97-113.
- Barcena-Ruiz, J.C. and M. Sedano (2011), "Weighted Welfare and Price Competition," *Japanese Economic Review* 62, pp.485-503.
- Bloomfield, M.J. (2018), "Compensation Disclosures and the Weaponization of Executive Pay: Evidence from Revenue-Based Performance Evaluation," SSRN, 286209.
- Capuano, C. and G. De Feo (2010), "Privatization in Oligopoly: The Impact of the Shadow Cost of Public Funds," *Rivista Italiana Degli Economisti* 15, pp.175-208.
- Hamilton, J.H. and S.T. Slutsky (1990), "Endogenous Timing in Duopoly Games: Stackelberg or Cournot Equilibria," *Games and Economic Behavior* 2, pp.29-46.
- Haraguchi, J. and K. Hirose (2018), "Endogenous Timing in a Price-Setting Mixed Oligopoly," MPRA Paper, No.87285.
- Harsanyi, J.C. and R. Selten (1988), A General Theory of Equilibrium Selection in Games, Cambridge, The MIT Press.
- Heywood, J.S. and G. Ye (2009), "Privatization and Timing in a Mixed Oligopoly with both Foreign and Domestic Firms," *Australian Economic Papers* 48, pp.320-332.
- Hirose, K. and T. Matsumura (2019), "Comparing Welfare and Profit in Quantity and Price Competition within Stackelberg Mixed Duopolies," *Journal of Economics* 126, pp.75-93.
- Hoppe, H.C. (2000), "Second-Mover Advantages in the Strategic Adoption of New Technology under Uncertainty," *International Journal of Industrial Organization* 18, pp.315-338.
- Lee, D., K. Choi and K. Hwang (2017), "First-Mover and Second-Mover Advantages in a Bilateral Duopoly," *The Korean Economic Review* 33, pp.35-53.
- Lee, S.-H. and L. Xu (2018), "Endogenous Timing in Private and Mixed Duopolies with Emission Taxes," *Journal of Economics* 124, pp.175-201.

- Lu, Y. (2006), "Endogenous Timing in a Mixed Oligopoly with Foreign Competitors: The Linear Demand Case," *Journal of Economics* 88, pp.49-68.
- Matsumura, T. (1998), "Partial Privatization in Mixed Duopoly," *Journal of Public Economy* 70, pp.473-483.
- Matsumura, T. (2003), "Endogenous Role in Mixed Markets: A Two-Production-Period Model," *Southern Economic Journal* 70, pp.403-413.
- Matsumura, T. and A. Ogawa (2010), "On the Robustness of Private Leadership in Mixed Duopoly," *Australian Economic Papers* 49, pp.149-160.
- Pal, D. (1998), "Endogenous Timing in a Mixed Oligopoly," Economics Letters 61, pp.181-185.
- Pi, J., P. Zhang and X. Chen (2018), "An Investigation of Stackelberg Mixed Oligopoly with Advertising Competition," *The Manchester School* 86, pp.468-487.
- Satoh, E. (2015), "Kongokasen Sijyo ni okeru Kyokyu Kuiki Kisei to Syohisya Yojyo: Nihon no Gas Sijyo Deta ni Motozuku Teiryoteki Bunseki [Territorial Restrictions and Consumer Welfare in a Mixed Oligopoly: The Japanese Gas Supply Market]," *Keizai Bunseki* [*Economic Analysis*], Economic and Social Research Institute (ESRI), 189, pp.23-43.
- Tomaru, Y. and Z. Kiyono (2010), "Endogenous Timing in Mixed Duopoly with Increasing Marginal Costs," *Journal of Institutional and Theoretical Economics* 166, pp.591-613.